SECURITIES AND EXCHANGE COMMISSION

FORM 8-K

Current report filing

Filing Date: **2001-08-03** | Period of Report: **2001-07-26** SEC Accession No. 0000912057-01-526424

(HTML Version on secdatabase.com)

FILER

BEMIS CO INC

CIK:11199| IRS No.: 430178130 | State of Incorp.:MO | Fiscal Year End: 1231

Type: 8-K | Act: 34 | File No.: 001-05277 | Film No.: 1696760

SIC: 2670 Converted paper & paperboard prods (no contaners/boxes)

Mailing Address Business Address
222 S 9TH STREET SUITE 222 S 9TH ST STE 2300
MINNEAPOLIS MN 55402-4099

MINNEAPOLIS MN 55402-4099 6123763000

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

Current Report

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report–July 26, 2001 (Date of earliest event reported)

BEMIS COMPANY, INC.

(Exact name of Registrant as specified in its charter)

Commission File Number 1-5277

Missouri

(State or other jurisdiction of incorporation or organization)

43-0178130

(I.R.S. Employer Identification No.)

222 South 9th Street, Suite 2300, Minneapolis, Minnesota 55402-4099

(Address of principal executive offices)

Registrant's telephone number, including area code: (612) 376-3000

Item 5. Other Events.

On July 26, 2001, Bemis Company, Inc. issued a press release containing its financial results for the fiscal quarter ended June 30, 2001, a copy of which is attached as an exhibit to this report.

Item 7. Financial Statements and Exhibits.

(c) Exhibits

- Amendment No. 2 to Purchase Agreement between the Registrant, Viskase Companies, Inc. and the Viskase-related companies listed therein, dated as of May 18, 2001.
- 10(j) First Amendment, dated as of June 21, 2000, to the Fourth Amended and Restated Credit Agreement dated as of

August 2, 1999, among the Registrant, the various financial institutions named therein and Morgan Guaranty Trust Company of New York, as Agent. 10(k) Bemis Company, Inc. 2001 Stock Incentive Plan.(1) Bridge Credit Agreement, dated as of January 12, 2001, among the Registrant, Morgan Guaranty Trust Company of 10(1) New York, as Administrative Agent, Bank One NA, as Documentation Agent, and the Banks listed therein. Letter Agreement, dated March 14, 2001, Re: Addition of Bank Under Bridge Credit Agreement, among the Registrant, 10(m)Morgan Guaranty Trust Company of New York, as Administrative Agent and as a Bank, Bank One NA, as Documentation Agent and as a Bank, and the Banks listed therein. Computation of Ratio of Earnings to Fixed Charges for fiscal quarters ended March 31, 2000 and 2001 and fiscal years 12 ended December 31, 1996, 1997, 1998, 1999 and 2000. 99 Press Release dated July 26, 2001. **(1)** Incorporated by reference to Exhibit B to the Registrant's Definitive Proxy Statement filed with the Securities and Exchange Commission on March 19, 2001 (File No. 1-5277).

2

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

BEMIS COMPANY, INC.

Ву	/s/ BENJAMIN R. FIELD, III Benjamin R. Field, III, Senior Vice President, Chief Financial Officer and Treasurer	Ву	/s/ GENE C. WULF Gene C. Wulf, Vice President and Controller
Date August 2, 2001		Date August 2, 2001	
		3	

Index to Exhibits

Exhibit Number	Description	Method of Filing
2(b)	Amendment No. 2 to Purchase Agreement between the Registrant, Viskase Companies, Inc. and the Viskase-related companies listed therein, dated as of May 18, 2001	Electronic Transmission
10(j)	First Amendment, dated as of June 21, 2000, to the Fourth	Electronic Transmission

therein and Morgan Guaranty Trust Company of New York, as Agent 10(k) Bemis Company, Inc. 2001 Stock Incentive Plan.(1) Incorporated by Reference Bridge Credit Agreement, dated as of January 12, 2001, among the Registrant, Morgan Guaranty Trust Company of New York, as 10(1) **Electronic Transmission** Administrative Agent, Bank One NA, as Documentation Agent, and the Banks listed therein Letter Agreement, dated March 14, 2001, Re: Addition of Bank Under Bridge Credit Agreement, among the Registrant, Morgan 10(m)Guaranty Trust Company of New York, as Administrative Agent **Electronic Transmission** and as a Bank, Bank One NA, as Documentation Agent and as a Bank, and the Banks listed therein

Computation of Ratio of Earnings to Fixed Charges for fiscal quarters ended March 31, 2000 and 2001 and fiscal years ended December 31, 1996, 1997, 1998, 1999 and 2000

Amended and Restated Credit Agreement dated as of August 2, 1999, among the Registrant, the various financial institutions named

Electronic Transmission

99 Press Release dated July 26, 2001

Electronic Transmission

(1)

Incorporated by reference to Exhibit B to the Registrant's Definitive Proxy Statement filed with the Securities and Exchange Commission on March 19, 2001 (File No. 1-5277).

4

QuickLinks

SIGNATURES
Index to Exhibits

AMENDMENT NO. 2 TO PURCHASE AGREEMENT

THIS AMENDMENT NO. 2, dated as of May 18, 2001 (this "Amendment"), to Purchase Agreement, dated as of July 7, 2000 (the "Purchase Agreement"), by and among Viskase Companies, Inc., a Delaware corporation ("Parent"), Viskase Corporation, a Pennsylvania corporation ("Viskase"), Viskase Holding Corporation, a Delaware corporation ("US Holdings"), Viskase Sales Corporation, a Delaware corporation ("Sales"), Viskase Europe Limited, a company organized under the laws of the United Kingdom ("Europe"), Viskase S.A., a company organized under the laws of France ("Viskase France"), Viskase Limited, a company organized under the laws of the United Kingdom ("UK"), Viskase Canada Inc., a company organized under the laws of Ontario ("Canada"), and Viskase Ireland Limited, a company organized under the laws of Ireland ("Ireland") (Parent, Viskase, US Holdings, Sales, Europe, Viskase France, UK, Canada and Ireland are each referred to herein individually as "Seller" and collectively as "Sellers"), and Bemis Company, Inc., a Missouri corporation ("Buyer").

WITNESETH:

WHEREAS, Buyer and Sellers have previously entered into the Purchase Agreement; and

WHEREAS, in order to set forth certain mutual agreements with respect to the Purchase Agreement and to resolve certain disagreements with respect to the definitive Valuation Date Statement, the parties desire to enter into this Amendment.

NOW, THEREFORE, in consideration of the premises and the mutual covenants contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE I

DEFINITIONS

- **1.1.** *Definitions.* In this Amendment, unless the context shall otherwise require, a term defined in the Purchase Agreement has the same meaning when used in this Amendment and a term defined anywhere in this Amendment has the same meaning throughout.
- 1.2. Interpretation. Each definition in this Amendment includes the singular and the plural, and reference to the neuter gender includes the masculine and feminine where appropriate. References to any statute or regulation means such statute or regulation as amended at the time and include any successor legislation or regulations. The heading to the Articles and Sections are for convenience of reference and shall not affect the meaning or interpretation of this Amendment. Except as otherwise stated, reference to Articles, Sections and Schedules mean the Articles, Sections and Schedules of this Amendment.

ARTICLE II

EMPLOYEE MATTERS

2.1. *Workers' Compensation.* (a) The following provisions shall be added to the end of Section 8.3(i) of the Purchase Agreement:

Sellers will remain responsible for all workers' compensation claims made by Affected Employees based on an injury occurring prior to the Closing Date, provided that (i) a written record of such injury existed in the employee's files prior to May 14, 2001 or the employee filed a claim regarding such injury with an appropriate Governmental Body prior to May 14, 2001, (ii) no aggravation of the injury has occurred on or after the Closing Date and (iii) the claim is approved on or prior to December 31, 2001. Except as provided in the preceding sentence, Buyer will be responsible for all workers' compensation claims made by Affected Employees. It is understood that Sellers will remain responsible for all workers' compensation claims made by employees of the Business (other than Affected Employees) based on an injury occurring prior to the Closing Date.

- (b) The parties agree that, as a result of Section 2.1(a) of this Amendment, (i) the original accrual of \$179,599 by Parent in its letter dated November 9, 2000, (ii) the revised accrual of \$259,000 by Buyer in its letter dated January 5, 2001 and (iii) the additional accrual of \$364,000 by Buyer in its letter dated January 5, 2001 are not necessary and will be eliminated in determining the definitive Valuation Date Statement pursuant to Section 3.2 of the Purchase Agreement. The parties hereby further agree that the definitive Valuation Date Statement will not include any liability, reserve or accrual for workers' compensation claims.
- **2.2.** *Employee Claims in Brazil.* (a) Schedule 2.3(d) to the Purchase Agreement shall be deleted in its entirety and replaced by Schedule 2.3(d) to this Amendment.
- (b) Schedule 2.4 to the Purchase Agreement shall be amended by adding the claims and proceedings listed in Schedule 2.4 to this Amendment.
- (c) The parties agree that, as a result of Sections 2.2(a) and (b) of this Amendment, the accrual of \$757,000 by Buyer in its letter dated January 5, 2001 is not necessary and will be eliminated in determining the definitive Valuation Date Statement pursuant to Section 3.2 of the Purchase Agreement. Buyer further agrees that the definitive Valuation Date Statement will not include any liability, reserve or accrual related to claims and/or lawsuits by employees in Brazil.

ARTICLE III

MISCELLANEOUS PROVISIONS

3.1. *Counterparts.* This Amendment may be executed in one or more counterparts, each of which will be deemed an original, but all of which together will constitute one and the same instrument.

2

- **3.2.** *Entire Agreement.* With respect to the subject matter hereof, this Amendment shall supersede anything to the contrary contained in the Purchase Agreement.
- **3.3.** *Partial Invalidity.* Wherever possible, each provision hereof shall be interpreted in such manner as to be effective and valid under applicable law, but in case any one or more of the provisions contained herein shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, such provision shall be ineffective to the extent, but only to the extent, of such invalidity, illegality or unenforceability without invalidating the remainder of such invalid, illegal or unenforceable provision or provisions or any other provisions hereof, unless such a construction would be unreasonable.
- **3.4.** *Governing Law.* This Amendment shall be governed by and construed in accordance with the internal laws (as opposed to the conflicts of law provisions) of the State of Illinois (without regard to the conflicts of law provisions thereof), except to the extent that the application of substantive laws of the United States or another jurisdiction is mandatory.
- **3.5.** *Waiver.* Any term or provision of this Amendment may be waived, or the time for its performance may be extended, by the party or parties entitled to the benefit thereof. Any such waiver shall be validly and sufficiently authorized for the purposes of this Amendment if, as to any party, it is authorized in writing by an authorized representative of such party. The failure of any party hereto to enforce at any time any

provision of this Amendment shall not be construed to be a waiver of such provision, nor in any way to affect the validity of this Amendment or any part hereof or the right of any party thereafter to enforce each and every such provision. No waiver of any breach of this Amendment shall be held to constitute a waiver of any other or subsequent breach.

3

IN WITNESS WHEREOF, this Amendment has been duly executed and delivered by the duly authorized officers of the parties hereto as of the date first above written.

VISKASE COMPANIES, INC.

/s/ GORDON S. DONOVAN

By: Name: Gordon S. Donovan

Title: Vice President

VISKASE CORPORATION

/s/ GORDON S. DONOVAN

By: Name: Gordon S. Donovan

Title: Vice President

VISKASE HOLDING CORPORATION

/s/ GORDON S. DONOVAN

By: Name: Gordon S. Donovan

Title: Vice President

VISKASE SALES CORPORATION

/s/ GORDON S. DONOVAN

By: Name: Gordon S. Donovan

Title: Vice President

VISKASE EUROPE LIMITED

/s/ GORDON S. DONOVAN

By: Name: Gordon S. Donovan

Title: Vice President

VISKASE S.A.

/s/ GORDON S. DONOVAN

By: Name: Gordon S. Donovan

Title: Vice President

4

VISKASE LIMITED

/s/ GORDON S. DONOVAN

By: Name: Gordon S. Donovan

Title: Vice President

VISKASE CANADA INC.

/s/ GORDON S. DONOVAN

By: Name: Gordon S. Donovan

Title: Vice President

VISKASE IRELAND LIMITED

/s/ GORDON S. DONOVAN

Name: Gordon S. Donovan

Title: Vice President

BEMIS COMPANY, INC.

/s/ SCOTT W. JOHNSON

By: Name: Scott W. Johnson

Title: Senior Vice President, General Counsel and

Secretary

5

By:

QuickLinks

AMENDMENT NO. 2 TO PURCHASE AGREEMENT

WITNESETH

ARTICLE I DEFINITIONS

ARTICLE II EMPLOYEE MATTERS

ARTICLE III MISCELLANEOUS PROVISIONS

FIRST AMENDMENT

THIS FIRST AMENDMENT dated as of June 21, 2000 (this "Amendment") amends the Fourth Amended and Restated Credit Agreement dated as of August 2, 1999 (the "Credit Agreement") among BEMIS COMPANY, INC., various financial institutions and MORGAN GUARANTY TRUST COMPANY OF NEW YORK, as Agent. Terms defined in the Credit Agreement are, unless otherwise defined herein or the context otherwise requires, used herein as defined therein.

WHEREAS, the Borrower, the Banks and the Agent have entered into the Credit Agreement; and

WHEREAS, the parties hereto desire to amend the Credit Agreement as set forth herein;

NOW, THEREFORE, the parties hereto agree as follows:

SECTION 1 *Amendments*. Subject to the effectiveness hereof pursuant to *Section 3*, the Credit Agreement shall be amended in accordance with *Sections 1.1* through *1.5* below.

- 1.1 Amendment to Definition of "Termination Date". The definition of "Termination Date" is amended by deleting the reference to "August 1, 2004" therein and substituting "August 1, 2005". The parties hereto agree that the foregoing amendment shall constitute an extension of the Termination Date pursuant to Section 2.14 of the Credit Agreement.
 - 1.2 Addition of New Definitions. The following new definitions are added to Section 1.1 in appropriate alphabetical order:

"Consolidated Debt" means at any date the consolidated Debt of the Borrower and its Consolidated Subsidiaries.

"Consolidated Net Worth" means at any date the consolidated stockholders' equity of the Borrower and its Consolidated Subsidiaries.

"Total Capital" means at any date the sum of (i) Consolidated Debt plus (ii) deferred taxes plus (iii) Consolidated Net Worth.

- 1.3 Deletion of Definition. The definition of "Consolidated Tangible Net Worth" is deleted.
- 1.4 Amendment of Definition of Debt. The definition of Debt is amended in its entirety to read as follows:

"Debt" of any Person means at any date, without duplication, (i) all obligations of such Person for borrowed money, (ii) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments, (iii) all obligations of such Person to pay the deferred purchase price of property or services, except trade accounts payable arising in the ordinary course of business, (iv) all obligations of such Person as lessee under capital leases, (v) all obligations of such Person to reimburse or indemnify the issuer of a letter of credit or Guarantee for drawings or payments thereunder, (vi) all Debt of others secured by a Lien on any asset of such Person, whether or not such Debt is assumed by such Person, and (vii) all Debt of others Guaranteed by such Person.

- 1.5 Amendment to Section 6.9. Section 6.9 is amended in its entirety to read as follows:
 - 6.9 Maximum Consolidated Debt to Total Capital Ratio. The Borrower will not permit the ratio of Consolidated Debt to Total Capital (expressed as a percentage) at any time to exceed (i) 55% prior to June 30, 2002 and (ii) 50% thereafter.

1.6 Amendment to Section 6.10. Section 6.10 is amended in its entirety to read as follows:

6.10 *Minimum Consolidated Net Worth.* The Borrower will not permit Consolidated Net Worth at any time to be less than the sum of (i) \$615,000,000 *plus* (ii) 50% of the consolidated net income of the Borrower and its Consolidated Subsidiaries in each completed fiscal quarter of the Borrower ending after March 31, 2000 (with no deduction for a net loss in any such fiscal quarter).

SECTION 2 *Representations and Warranties*. The Borrower represents and warrants to the Banks and the Agent that (a) each of the representations and warranties set forth in Section 5 of the Credit Agreement is true and correct as of the date hereof, with the same effect as if made on such date (except to the extent such representations and warranties expressly refer to an earlier date, in which case they were true and correct as of such earlier date), (b) the execution and delivery hereof by the Borrower and the performance by the Borrower of its obligations under the Credit Agreement as amended hereby (as so amended, the "Amended Credit Agreement") (i) are within the corporate powers of the Borrower, (ii) have been duly authorized by all necessary action on the part of the Borrower, (iii) have received all necessary governmental approval and (iv) do not and will not contravene or conflict with (x) any provision of applicable law or the certificate of incorporation or by-laws or other organizational documents of the Borrower or (y) any agreement, judgement, injunction, order, decree or other instrument binding upon the Borrower and (c) the Amended Credit Agreement is a legal, valid and binding obligation of the Borrower enforceable against the Borrower in accordance with its terms, except as enforceability may be limited by bankruptcy, insolvency or similar laws of general application affecting the enforcement of creditors' rights or by general principles of equity limiting the availability of equitable remedies.

- 2-

SECTION 3 *Effectiveness*. (a) The amendment set forth in *Section 1.1* above shall become effective when all Banks have signed counterparts of this Amendment consenting to the extension of the Termination Date accomplished thereby and the Agent shall have received such counterparts (by facsimile or otherwise) and a counterpart hereof executed by the Company.

(b) The amendments set forth in *Sections 1.2* through *1.6* above shall become effective when the Required Banks have signed counterparts of this Amendment consenting to such amendments and the Agent shall have received such counterparts (by facsimile or otherwise) and a counterpart hereof executed by the Company.

SECTION 4 Miscellaneous.

- 4.1 *Counterparts*. This Amendment may be executed in any number of counterparts and by the different parties on separate counterparts, and each such counterpart shall be deemed to be an original but all such counterparts shall together constitute one and the same Amendment.
- 4.2 *Expenses*. The Borrower agrees to pay all reasonable expenses of the Agent, including reasonable fees and charges of special counsel to the Agent, in connection with the preparation, execution and delivery of this Amendment.
 - 4.3 Governing Law. This Amendment shall be construed in accordance with and governed by the law of the State of New York.
- 4.4 Successors and Assigns. This Amendment shall be binding upon the Borrower, the Banks and the Agent and their respective successors and assigns, and shall inure to the benefit of the Borrower, the Banks and the Agent and the respective successors and assigns of the Banks and the Agent.

IN WITNESS WHEREOF, the parties have caused this Amendment to be duly executed and delivered by their proper and duly authorized officers as of the day and year first above written.

BEMIS COMPANY, INC.

By:

Title:

- 3-

	MORGAN GUARANTY TRUST COMPANY OF NEW YORK, as Agent
	By: Title:
Each of the undersigned Banks consents to the exte	ension of the Termination Date pursuant to Section 1.1 of the foregoing Amendment:
	BANK ONE, NA (Main Office Chicago) (formerly The First National Bank of Chicago)
	By: Title:
	NORWEST BANK MINNESOTA, N.A.
	By: Title:
	U.S. BANK NATIONAL ASSOCIATION
	By: Title:
	WACHOVIA BANK, N.A.
	By: Title:
	-4-
	FIRST HAWAIIAN BANK
	By: Title:
	REVOLVING COMMITMENT VEHICLE CORPORATION
	Morgan Guaranty Trust Company of New York, By: as Attorney-in-fact for Revolving Commitment Vehicle Corporation
	By:

	Title:
Each of the undersigned Banks consents to the amendments s	et forth in Sections 1.2 through 1.6 of the foregoing Amendment.
	K ONE, NA (Main Office Chicago) verly The First National Bank of Chicago)
By:	Title:
NOR	WEST BANK MINNESOTA, N.A.
By:	Title:
	-5-
U.S. I	BANK NATIONAL ASSOCIATION
By:	Title:
WAC	CHOVIA BANK, N.A.
By:	Title:
FIRS	T HAWAIIAN BANK
By:	Title:
	OLVING COMMITMENT VEHICLE PORATION
Ву:	Morgan Guaranty Trust Company of New York, as Attorney-in-fact for Revolving Commitment Vehicle Corporation
By:	Title:
	- 6-

QuickLinks

FIRST AMENDMENT

BRIDGE CREDIT AGREEMENT

Dated as of January 12, 2001

AMONG

BEMIS COMPANY, INC.

MORGAN GUARANTY TRUST COMPANY

OF NEW YORK, AS ADMINISTRATIVE AGENT,

BANK ONE, NA, AS DOCUMENTATION AGENT,

AND

THE BANKS LISTED HEREIN

\$230,000,000

<Page>

<Table> <Caption>

TABLE OF CONTENTS

SECTION			PAGE
<s></s>			<c></c>
SECTION	1.	INTERPRETATIONS AND DEFINITIONS	1
	1.1	Definitions	1
	1.2	Other Definitional Provisions	8
SECTION	2.	THE LOANS	9
	2.1	The Loans	
	2.2	Limitations on Borrowings and Interest Periods; Loans Pro Rata	9
	2.3	Method of Borrowing	10
	2.4	The Notes	11
	2.5	Maturity of Loans	
	2.6	Interest Rates	11
	2.7	Fees	
	2.8	Optional Termination or Reduction of Commitments	
	2.9	Mandatory Termination or Reduction of Commitments	
	2.10	Voluntary Prepayments	
	2.12	General Provisions as to Payments	
	2.13	Computation of Interest and Fees	
	2.14	Funding Losses	17
SECTION	3.	CONDITIONS OF LENDING	17
	3.1	All Loans	
	3.2	Initial Loans	17
SECTION	4.	CHANGE IN CIRCUMSTANCES AFFECTING FIXED RATE LOANS	18

	4.1	Basis for Determining Interest Rate Inadequate18
	4.2	Illegality
	4.3	Increased Costs and Rate of Return
SECTION	5.	REPRESENTATIONS AND WARRANTIES
	5.1	Corporate Existence and Power
	5.2	Corporate Authorization
	5.3	Binding Effect
	5.4	Financial Statements
	5.5	Litigation
	5.6	Taxes
	5.7	Governmental and other Approvals
	5.8	Compliance with ERISA
	5.9	Environmental Matters
CECHION	6	COVENANTS
SECTION	6.	COVENANTS23
		i
<page></page>		
	6.1	Financial Statements
	6.2	Maintenance of Existence
	6.3	Maintenance of Properties25
	6.4	Compliance with Laws
	6.5	Notice of Proceedings
	6.6	Use of Proceeds
	6.7	Payment of Taxes
	6.8	Insurance
	6.9	Maximum Consolidated Debt to Total Capital Ratio26
	6.10	Minimum Consolidated Net Worth26
	6.11	Negative Pledge
	6.12	Consolidations, Mergers and Sales of Assets27
SECTION	7.	EVENTS OF DEFAULT
SECTION	8.	THE ADMINISTRATIVE AGENT29
	8.1	Appointment and Authorization
	8.2	Administrative Agent and Affiliates29
	8.3	Action by Administrative Agent
	8.4	Consultation with Experts29
	8.5	Liability of Administrative Agent
	8.6	Indemnification
	8.7	Failure to Act
	8.8	Resignation or Removal of Administrative Agent
	8.9	Credit Decision
SECTION	9.	MISCELLANEOUS31
	9.1	Notices
	9.2	Amendments and Waivers; Cumulative Remedies
	9.3	Successors and Assigns32
	9.4	Indemnification by the Borrower; Documentary Taxes
	9.5	Sharing of Set-Offs35
	9.6	Collateral
	9.7	Counterparts35
	9.8	Headings; Table of Contents
	9.9	Governing Law
∠ /m - 1- 1 · •	9.10	Documentation Agent

 • | || EXHIBIT | | Form of Note |
| EXHIBIT | | Form of Opinion of General Counsel |
| EXHIBIT C | | Form of Opinion of Mayer. Brown & Platt |
Form of Assignment and Assumption Agreement EXHIBIT D

Pricing Schedule

ii

<Page>

BRIDGE CREDIT AGREEMENT (this "Agreement") dated as of January 12, 2001 among BEMIS COMPANY, INC., a Missouri corporation (the "Borrower"), the BANKS listed on the signature pages hereof (together with their respective successors and assigns, the "Banks") and MORGAN GUARANTY TRUST COMPANY OF NEW YORK, a New York State banking corporation, as administrative agent for the Banks (the "Administrative Agent").

The parties hereto agree as follows:

SECTION 1. INTERPRETATIONS AND DEFINITIONS

1.1 DEFINITIONS. The following terms, as used herein, shall have the following respective meanings:

"Adjusted CD Rate" has the meaning set forth in Section 2.6(b).

"Adjusted Euro-Dollar Rate" has the meaning set forth in Section 2.6(c).

"Administrative Agent" has the meaning set forth in the introductory paragraph.

"Administrative Questionnaire" means, for any Bank, the administrative questionnaire submitted by the Bank to the Administrative Agent that sets forth certain administrative details requested by the Administrative Agent.

"Approved Fund" means any Fund that is administered or managed by (a) a Bank, (b) an affiliate of a Bank or (c) an entity or an affiliate of an entity that administers or manages a Bank.

"Assessment Rate" has the meaning set forth in Section 2.6(b).

"Assignment and Acceptance" means an assignment and acceptance agreement, substantially in the form of Exhibit D hereto.

"Banks" has the meaning set forth in the introductory paragraph.

"Base Rate" means, for any day, a rate per annum equal to the higher of (i) the Prime Rate for such day and (ii) the sum of 1/2 of 1% plus the Federal Funds Rate for such day.

"Base Rate Loan" means a Loan which the Borrower specifies pursuant to Section $2.3\,$ shall be a Base Rate Loan.

<Page>

"Borrower" has the meaning set forth in the introductory paragraph.

"CD Base Rate" has the meaning set forth in Section 2.6(b).

"CD Loan" means a Loan which the Borrower specifies pursuant to Section $2.3 \ \mathrm{shall} \ \mathrm{be} \ \mathrm{a} \ \mathrm{CD} \ \mathrm{Loan}.$

"CD Margin" has the meaning set forth in section 2.6(b).

"CD Reserve Percentage" has the meaning set forth in Section 2.6(b).

"Change of Control" means the occurrence of any of the following events: (x) any "person" or "group" (within the meaning of Section 13 or 14 of the Securities Exchange Act of 1934 (the "Exchange Act") becomes the beneficial owner (as defined in Rule 13d-3 under the Exchange Act) of 20% or more of the fully diluted Voting Securities of the Borrower or (y) individuals who at the beginning of any period of two consecutive calendar years constituted the board of directors of the Borrower (together with any new directors whose election by the board of directors of the Borrower or whose nomination for election by the Borrower's shareholders was approved by the members of the board of directors of the Borrower at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason to constitute a majority of the members of the board of directors of the Borrower.

"Code" means the Internal Revenue Code of 1986.

"Commitment" means, with respect to each Bank, the amount set forth opposite the name of such Bank on the signature pages hereof, as such amount may be changed from time to time pursuant to the terms hereof.

"Consolidated Debt" means at any date the consolidated Debt of the Borrower and its Consolidated Subsidiaries.

"Consolidated Net Worth" means at any date the consolidated stockholders' equity of the Borrower and its Consolidated Subsidiaries.

"Consolidated Subsidiary" means at any date any Subsidiary or other entity the accounts of which would be consolidated with those of the Borrower in its consolidated financial statements as of such date.

"Controlled Group" means all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control which, together with the Borrower, are treated as a single employer under Section 414(b) or 414(c) of the Code.

2

<Page>

"Debt" of any Person means at any date, without duplication, (i) all obligations of such Person for borrowed money, (ii) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments, (iii) all obligations of such Person to pay the deferred purchase price of property or services, except trade accounts payable arising in the ordinary course of business, (iv) all obligations of such Person as lessee under capital leases, (v) all obligations of such Person to reimburse or indemnify the issuer of a letter of credit or Guarantee for drawings or payments thereunder, (vi) all Debt of others secured by a Lien on any asset of such Person, whether or not such Debt is assumed by such Person, and (vii) all Debt of others Guaranteed by such Person

"Default" means any event or condition which constitutes an Event of Default or which with the giving of notice or lapse of time, or both, would become an Event of Default.

"Dollars" and the sign \$\$ mean lawful money of the United States of America.

"Domestic Business Day" means any day except a Saturday, Sunday or other day on which commercial banks in New York City are authorized by law to close.

"Domestic Lending Office" means, as to each Bank, its office located at its address set forth on the signature pages hereof (or identified on the signature pages hereof as its Domestic Lending Office) or such other office as such Bank may hereafter designate as its Domestic Lending Office by notice to the Borrower and the Administrative Agent; PROVIDED that any Bank may from time to time by notice to the Borrower and the Administrative Agent designate separate Domestic Lending Offices for its Base Rate Loans, on the one hand, and its CD Loans, on the other hand, in which case all references herein to the Domestic Lending Office of such Bank shall be deemed to refer to either or both of such offices, as the context may require.

"Domestic Loans" means CD Loans or Base Rate Loans or both.

"Eligible Assignee" means (a) a Bank, (b) an affiliate of a Bank, (c) an Approved Fund, and (d) any other Person (other than a natural Person) approved by the Administrative Agent and, unless (x) such Person is taking delivery of an assignment in connection with physical settlement of a credit derivatives transaction or (y) an Event of Default has occurred and is continuing, the Borrower (each such approval not to be unreasonably withheld or delayed). If the consent of the Borrower to an assignment or to an Eligible Assignee is required hereunder (including a consent to an assignment which does not meet the minimum assignment thresholds specified in paragraph (b) (i) of Section 9.3), the Borrower shall be deemed to have given its consent five

Domestic Business Days after the date notice thereof has been delivered by the assigning Bank (through the Administrative Agent) unless such consent is expressly refused by the Borrower prior to such fifth Domestic Business Day.

"Environmental Laws" means any and all federal, state, local and foreign statutes, laws, judicial decisions, regulations, ordinances, rules, judgments, orders, decrees,

3

<Page>

injunctions, permits, concessions, grants, franchises, licenses, agreements and other governmental restrictions relating to the environment, to the effect of the environment on human health or to emissions, discharges or releases of pollutants, contaminants, Hazardous Substances or wastes into the environment, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of pollutants, contaminants, Hazardous Substances or wastes or the clean-up or other remediation thereof.

"ERISA" means the Employee Retirement Income Security Act of 1974.

"Euro-Dollar Business Day" means any Domestic Business Day on which commercial banks in London are open for domestic and international business (including dealings in Dollar deposits).

"Euro-Dollar Lending Office" means, as to each Bank, its office or branch located at its address set forth on the signature pages hereof (or identified on the signature pages hereof as its Euro-Dollar Lending Office) or such other branch (or affiliate) of such Bank as it may hereafter designate as its Euro-Dollar Lending Office by notice to the Borrower and the Administrative Agent.

"Euro-Dollar Loan" means a Loan which the Borrower specifies pursuant to Section $2.3 \ \text{shall}$ be a Euro-Dollar Loan.

"Euro-Dollar Margin" has the meaning set forth in Section 2.6(c).

"Euro-Dollar Reserve Percentage" has the meaning set forth in Section 2.6(c).

"Event of Default" has the meaning set forth in Section 7.

"Existing Credit Agreement" means the Fourth Amended and Restated Credit Agreement dated as of August 2, 1999 among the Borrower, various financial institutions and Morgan Guaranty Trust Company of New York, as agent.

"Federal Funds Rate" means, for any day, the rate per annum (rounded upward, if necessary, to the nearest 1/100th of 1%) equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Domestic Business Day next succeeding such day; PROVIDED that (i) if such day is not a Domestic Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Domestic Business Day as so published on the next succeeding Domestic Business Day, and (ii) if no such rate is so published on such next succeeding Domestic Business Day, the Federal Funds Rate for such day shall be the average rate quoted to Morgan Guaranty Trust Company of New York on such day on such transactions as determined by the Administrative Agent.

4

<Page>

"Fixed CD Rate" has the meaning set forth in Section 2.6(b).

"Fixed Rate Loans" means CD Loans or Euro-Dollar Loans or both.

"Fund" means any Person (other than a natural Person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"Guarantee" by any Person means any obligation, contingent or otherwise, of such Person directly or indirectly guaranteeing any Debt or other

obligation of any other Person or in any manner providing for the payment of any Debt of any other Person or otherwise protecting the holder of such Debt against loss (whether by agreement to keep-well, to purchase assets, goods, securities or services, to take-or-pay, or to maintain financial statement conditions or otherwise); PROVIDED that the term Guarantee shall not include endorsements for collection or deposit in the ordinary course of business. The term "Guarantee" used as a verb has a corresponding meaning.

"Hazardous Substances" means any toxic, radioactive, caustic or otherwise hazardous substance, including petroleum, its derivatives, by-products and other hydrocarbons, or any substance having any constituent elements displaying any of the foregoing characteristics.

"Interest Period" means:

- (1) with respect to each CD Loan, the period commencing on the date of such Loan and ending 30, 60, 90 or 180 days thereafter, as the Borrower may elect; PROVIDED that any Interest Period which would otherwise end on a day which is not a Euro-Dollar Business Day shall be extended to the next succeeding Euro-Dollar Business Day;
- (2) with respect to each Euro-Dollar Loan, the period commencing on the date of such Loan and ending one, two, three or six months thereafter, as the Borrower may elect; PROVIDED that (a) any Interest Period which would otherwise end on a day which is not a Euro-Dollar Business Day shall be extended to the next succeeding Euro-Dollar Business Day unless such Euro-Dollar Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Euro-Dollar Business Day and (b) any Interest Period which begins on the last Euro-Dollar Business Day of the calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Euro-Dollar Business Day of a calendar month; and
- (3) with respect to each Base Rate Loan, the period commencing on the date of such Loan and ending 30 days thereafter.

5

<Page>

Any Interest Period which begins before the Termination Date and would otherwise end after the Termination Date shall end on the Termination Date.

"Lien" means, with respect to any asset, (i) any lien, charge, mortgage, security interest, pledge or other encumbrance of any kind in respect of such asset or (ii) the interest of a vendor or lessor under any conditional sale agreement, capital lease or other title retention agreement relating to such asset.

"Loan" and "Loans" means a Domestic Loan or a Euro-Dollar Loan, or both, as the context may require.

"London Interbank Offered Rate" has the meaning set forth in Section 2.6(c).

"Material Subsidiary" means at any time a Subsidiary which as of such time meets the definition of a "significant subsidiary" contained as of the date hereof in Regulation S-X of the Securities and Exchange Commission.

"Moody's" means Moody's Investors Service, Inc.

"Net Cash Proceeds" means, with respect to any event, (a) the cash proceeds received in respect of such event including (i) any cash received in respect of any non-cash proceeds, but only as and when received, and (ii) in the case of a condemnation or similar event, condemnation awards and similar payments, net of (b) the sum of (i) all reasonable fees and out-of-pocket expenses paid by the Borrower and its Subsidiaries to third parties in connection with such event, (ii) in the case of a sale, transfer or other disposition of an asset (including pursuant to a sale and leaseback transaction or a condemnation or similar proceeding), the amount of all payments required to be made by the Borrower and its Subsidiaries as a result of such event to repay Debt (other than Loans) secured by such asset or otherwise subject to mandatory prepayment as a result of such event, and (iii) the amount of all taxes paid (or

reasonably estimated to be payable) by the Borrower and its Subsidiaries, and the amount of any reserves established by the Borrower and its Subsidiaries to fund contingent liabilities reasonably estimated to be payable, in each case during the year that such event occurred or the next succeeding year and that are directly attributable to such event (as determined reasonably and in good faith by the chief financial officer of the Borrower).

"Note" means a promissory note of the Borrower, substantially in the form of Exhibit A hereto, evidencing the obligation of the Borrower to repay Loans.

"Participant" has the meaning set forth in Section 9.3(d).

"PBGC" means the Pension Benefit Guaranty Corporation or any entity succeeding to any or all of its functions under ERISA.

6

<Page>

"Person" means an individual, a corporation, a partnership, a limited liability company, an association, a business trust or any other entity or organization, including a government or political subdivision or an agency or instrumentality thereof.

"Plan" means at any time an employee pension benefit plan which is covered by Title IV of ERISA or subject to the minimum funding standards under Section 412 of the Code and is either (i) maintained by a member of the Controlled Group for employees of a member of the Controlled Group or (ii) maintained pursuant to a collective bargaining agreement or any other arrangement under which more than one employer makes contributions and to which a member of the Controlled Group is then making or accruing an obligation to make contributions or has within the preceding five plan years made contributions.

"Pricing Schedule" means the Pricing Schedule attached hereto.

"Prime Rate" means the rate of interest publicly announced by Morgan Guaranty Trust Company of New York in New York City from time to time as its Prime Rate.

"Reduction Event" means:

- (a) any sale, lease or other disposition (including any such transaction effected by way of merger or consolidation) by the Borrower or any of its Subsidiaries of any asset, including without limitation any sale-leaseback transaction, whether or not involving a capital lease, but excluding (i) dispositions of temporary cash investments, inventory and used, surplus or worn out equipment in the ordinary course of business and (ii) dispositions to the Borrower or any Subsidiary of the Borrower; PROVIDED that a disposition of assets not excluded by clauses (i) and (ii) above shall not constitute a Reduction Event unless and until (and only to the extent that) the aggregate Net Cash Proceeds from such disposition, when combined with all other such dispositions previously made after the date hereof, exceeds \$50,000,000; or
- (b) the issuance by the Borrower or any Subsidiary of any equity securities, or the receipt by the Borrower or any Subsidiary of any capital contribution, other than (i) any such issuance of equity securities to, or receipt of any such capital contribution from, the Borrower or a Subsidiary and (ii) the issuance by the Borrower of its common stock pursuant to employee benefit plans in the ordinary course of business; or
- (c) the incurrence by the Borrower or any Subsidiary of any Debt after the date hereof, other than (i) Debt under this Agreement, (ii) Debt under the Existing Credit Agreement, (iii) the issuance by the Borrower of commercial paper backstopped by this Agreement or (iv) up to \$120,000,000 in aggregate principal amount of extendible commercial notes issued by the Borrower.

"Reference Bank" means Morgan Guaranty Trust Company of New York.

7

<Page>

"Refunding Loan" means a Loan which, after application of the proceeds

thereof, results in no net increase in the outstanding principal amount of Loans made by any Bank.

"Register" has the meaning set forth in Section 9.3(c).

"Regulation U" means Regulation U of the Board of Governors of the Federal Reserve System, as in effect from time to time.

"Required Banks" means at any time Banks holding at least 66-2/3% of the aggregate unpaid principal amount of the Notes or, if no Loans are at the time outstanding hereunder, Banks having at least 66-2/3% of the aggregate amount of the Commitments.

"S&P" means Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc.

"Subsidiary" means any corporation or other entity of which capital stock or other ownership interests having ordinary voting power to elect a majority of the board of directors or other persons performing similar functions is at the time directly or indirectly owned by the Borrower.

"Termination Date" means January 11, 2002.

"Total Capital" means at any date the sum of (i) Consolidated Debt PLUS (ii) deferred taxes PLUS (iii) Consolidated Net Worth.

"Unfunded Vested Liabilities" means, with respect to any Plan, the amount, if any, by which the present value of all vested benefits under such Plan exceeds the fair market value of all Plan assets allowable to such benefits, as determined on the most recent valuation date of such Plan, but only to the extent that excess represents a potential liability of the Borrower or any member of the Controlled Group to the PBGC or to such Plan under Title IV of ERISA.

"Voting Securities" means any securities having ordinary power to vote for the election of directors.

"Wholly-Owned Consolidated Subsidiary" means any Consolidated Subsidiary all of the shares of capital stock or other ownership interests of which (except directors' qualifying shares) are at the time directly or indirectly owned by the Borrower.

1.2 OTHER DEFINITIONAL PROVISIONS.

(a) Unless otherwise specified herein, all accounting terms used herein shall be interpreted, all accounting determinations hereunder shall be made, and all financial statements required to be delivered hereunder shall be prepared in

8

<Page>

accordance with generally accepted accounting principles as in effect from time to time, applied on a basis consistent with the most recent audited consolidated financial statements of the Borrower and its Consolidated Subsidiaries delivered to the Bank.

- (b) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.
- (c) SECTION, SCHEDULE and EXHIBIT references are to this Agreement unless otherwise specified.
- (d) The term "including" is not limiting and means "including without limitation."
- (e) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including"; the words "to" and "until" each mean "to but excluding", and the word "through" means "to and including."
- (f) Unless otherwise expressly provided herein,(i) references to agreements (including this Agreement) and other

contractual instruments shall be deemed to include all subsequent amendments and other modifications thereto, but only to the extent such amendments and other modifications are not prohibited by the terms of this Agreement or any Note, and (ii) references to any statute or regulation are to be construed as including all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting such statute or regulation.

SECTION 2. THE LOANS.

- 2.1 THE LOANS. From the date hereof, to and excluding the Termination Date, each Bank severally agrees, on the terms and conditions contained in this Agreement, to lend to the Borrower from time to time amounts not exceeding in the aggregate at any one time outstanding the amount of its Commitment. During such period and subject to the limitations set forth in Section 2.2, the Borrower may borrow under this Section 2.1, repay or prepay Loans and reborrow under this Section 2.1.
- 2.2 LIMITATIONS ON BORROWINGS AND INTEREST PERIODS; LOANS PRO RATA. Each borrowing of Loans hereunder shall be in the principal amount of \$10,000,000 or a higher integral multiple of \$1,000,000 (except that any borrowing may be in the aggregate amount of the unused Commitments) and shall be made by the several Banks ratably in proportion to their respective Commitments. No more than five Interest Periods with respect to Fixed Rate Loans may be outstanding at any time.

9

<Page>

2.3 METHOD OF BORROWING.

- (a) With respect to each borrowing of Loans made pursuant to Section 2.1, the Borrower shall give the Administrative Agent notice (a "Notice of Borrowing") not later than 10:30 a.m. (New York City time) on the same date of each borrowing of Base Rate Loans, at least one Domestic Business Day before each borrowing of CD Loans, or at least three Euro-Dollar Business Days before each borrowing of Euro-Dollar Loans, specifying:
 - (i) the date of such Loans, which shall be a Domestic Business Day in the case of Domestic Loans and a Euro-Dollar Business Day in the case of Euro-Dollar Loans;
 - $\hspace{1.5cm} \hbox{(ii)} \hspace{1.5cm} \hbox{the principal amount of such borrowings of Loans;} \\$
 - $\mbox{(iii)} \qquad \mbox{whether such Loans are to be Base Rate Loans, CD Loans or Euro-Dollar Loans; and} \\$
 - (iv) in the case of Fixed Rate Loans, the duration of the Interest Period applicable thereto, subject to the definition of Interest Period.
- (b) Upon receipt of a Notice of Borrowing, the Administrative Agent shall promptly notify each Bank of the contents thereof and of such Bank's ratable share of the Loan specified therein and such Notice of Borrowing shall not thereafter be revocable by the Borrower.
- (c) Not later than noon (New York City time) on the date of each borrowing of Loans, each Bank shall (except as provided in Section 2.3(d)) make available its ratable share of such borrowing, in Federal or other funds immediately available in New York City, to the Administrative Agent at its address set forth on the signature pages hereof or at such other address as it may hereafter designate by notice to the Borrower and the Banks and, unless the Administrative Agent determines that any applicable condition specified in Section 4 has not been satisfied, the Administrative Agent will promptly make the funds so received from the Banks available to the Borrower at the Administrative Agent's aforesaid address.
- (d) If any Bank makes a new Loan hereunder on a day on which the Borrower is to repay all or any part of an outstanding Loan

from such Bank, such Bank shall apply the proceeds of its new Loan to make such repayment and only an amount equal to the difference (if any) between the amount being borrowed and the amount being repaid shall be made available by such Bank to the Administrative Agent as provided in Section 2.3(c), or remitted by the Borrower to the Administrative Agent as provided in Section 2.11, as the case may be.

10

<Page>

- 2.4 THE NOTES. (a) The Loans of each Bank shall be evidenced by a single Note payable to the order of such Bank for the account of its applicable lending office.
 - (b) Each Bank may, by notice to the Borrower and the Administrative Agent, request that its Loans of a particular type be evidenced by a separate Note in an amount equal to the aggregate unpaid principal amount of such Loans. Each reference in this Agreement to the "Note" of such Bank shall be deemed to refer to and include any or all of such Notes, as the context may require.
 - (c) Upon receipt of each Bank's Note pursuant to Section 3.2(a), the Administrative Agent shall deliver such Note to such Bank. Each Bank shall record and, prior to any transfer of its Note, shall endorse on the schedules forming a part thereof appropriate notations evidencing the date, the amount and the maturity of each Loan to be evidenced by such Note and the date and amount of each payment of principal made by the Borrower with respect thereto; PROVIDED that failure to make any such endorsement or notation shall not affect the obligations of the Borrower hereunder or under any Note. Each Bank is hereby irrevocably authorized by the Borrower so to endorse the Notes and to attach to and make a part of any Note a continuation of any such schedule as and when required.
- 2.5 MATURITY OF LOANS. Each Loan shall mature, and the principal amount thereof shall be due and payable, on the last day of the Interest Period applicable to such Loan.

2.6 INTEREST RATES.

- (a) Each Base Rate Loan shall bear interest on the outstanding principal amount thereof, for each day from the date such Loan is made until it becomes due, at a rate per annum equal to the Base Rate. Such interest shall be payable for each Interest Period on the last day thereof. Overdue principal of and, to the extent permitted by law, overdue interest on the Base Rate Loans shall bear interest for each day until paid at a rate per annum equal to the sum of 1% plus the otherwise applicable rate for such day.
- (b) Each CD Loan shall bear interest on the outstanding principal amount thereof, for the Interest Period applicable thereto, at a rate per annum equal to the applicable Fixed CD Rate. Such interest shall be payable for each Interest Period on the last day thereof and, if such Interest Period is longer than 90 days, at intervals of 90 days after the first day thereof. Any overdue principal of and, to the extent permitted by law, overdue interest on the CD Loans shall bear interest for each day until paid at a rate per annum equal to the sum of 1% plus the rate applicable to Base Rate Loans for such day.

The "Fixed CD Rate" applicable to any CD Loan for any Interest Period means a rate per annum equal to the sum of the CD Margin plus the applicable Adjusted CD Rate.

11

<Page>

The "CD Margin" means a rate per annum determined in accordance with the Pricing Schedule.

The "Adjusted CD Rate" applicable to any Interest Period means a rate per annum determined pursuant to the following formula:

ACDR = CDBR + AR

[1 - CDRP]

ACDR = Adjusted CD Rate for such Interest Period

CDBR = CD Base Rate for such Interest Period

AR = Assessment Rate

CDRP = CD Reserve Percentage

The "CD Base Rate" means for any Interest Period the prevailing per annum rate of interest (rounded upward, if necessary, to the next higher (1/100 of 1%) determined by the Administrative Agent to be the average of bid rates quoted to the Reference Bank at 10:00 a.m. (New York City time) (or as soon thereafter as is practicable) on the first day of such Interest Period by two or more New York certificate of deposit dealers of recognized standing selected by the Reference Bank for the purchase at face value from the Reference Bank of its certificates of deposit in an amount comparable to the principal amount of the CD Loan made by the Reference Bank to which such Interest Period applies and with a maturity comparable to such Interest Period.

The "CD Reserve Percentage" means for any day, that percentage, expressed as a decimal, which is in effect on such day, as prescribed by the Board of Governors of the Federal Reserve System (or any successor) for determining the maximum reserve requirement (including any basic, supplemental or emergency reserves) for a member bank of the Federal Reserve System in New York City with deposits exceeding five billion Dollars in respect of new non-personal time deposits in Dollars in New York City having a maturity comparable to the related Interest Period and in an amount of \$100,000 or more. The Fixed CD Rate shall be adjusted automatically on and as of the effective date of any change in the CD Reserve Percentage.

The "Assessment Rate" for any Interest Period means a rate per annum (rounded upwards, if necessary, to the next higher 1/100 of 1%) determined by the Administrative Agent to be the rate per annum equal to (i) the gross annual assessment rate payable to the Federal Deposit Insurance Corporation (or any successor) by the Reference Bank for insuring the Reference Bank's domestic Dollar deposits less (ii) the most recently determined credit against such assessments, expressed as an annual rate, available under applicable law to the Reference Bank, in each case as determined by the Reference Bank as of the first day of such Interest Period.

12

<Page>

(c) Each Euro-Dollar Loan shall bear interest on the unpaid principal amount thereof, for the Interest Period applicable thereto, at a rate per annum equal to the sum of the Euro-Dollar Margin plus the applicable Adjusted Euro-Dollar Rate. Such interest shall be payable for each Interest Period on the last day thereof and, if such Interest Period is longer than three months, at intervals of three months after the first day thereof.

The "Adjusted Euro-Dollar Rate" applicable to any Interest Period means a rate per annum equal to the quotient obtained (rounded upwards, if necessary, to the next higher 1/100 of 1%) by dividing (i) the applicable London Interbank Offered Rate by (ii) 1.00 MINUS the Euro-Dollar Reserve Percentage.

The "London Interbank Offered Rate" applicable to any Interest Period means the rate determined by the Administrative Agent to be the average of the rates per annum at which deposits in Dollars are offered to the Reference Bank in the London interbank market at approximately 11:00 a.m. (London Time) two Euro-Dollar Business Days prior to the first day of such Interest Period in an amount approximately equal to the aggregate unpaid principal amount of the Euro-Dollar Loan made by the Reference Bank to which such Interest Period is to apply and for a period of time comparable to such Interest Period.

The "Euro-Dollar Reserve Percentage" means for any day that percentage (expressed as a decimal) which is in effect on such day, as prescribed by the Board of Governors of the Federal Reserve System (or any successor) for determining the maximum reserve requirement for a member bank of the Federal Reserve System in New York City with deposits exceeding five billion dollars in

respect of "Eurocurrency liabilities" (or in respect of any other category of liabilities which includes deposits by reference to which the interest rate on Euro-Dollar Loans is determined or any category of extensions of credit or other assets which includes loans by a non-United States office of any Bank to United States residents). The Adjusted Euro-Dollar Rate shall be adjusted automatically on and as of the effective date of any change in the Euro-Dollar Reserve Percentage.

The "Euro-Dollar Margin" means a rate per annum determined in accordance with the Pricing Schedule.

Any overdue principal of and, to the extent permitted by law, overdue interest on, any Euro-Dollar Loan shall bear interest payable on demand, for each day from the date payment thereof was due to the date of actual payment, at a rate per annum equal to 1% plus the Euro-Dollar Margin plus the quotient obtained (rounded upwards, if necessary, to the next higher 1/100 of 1%) by dividing (i) the interest rate per annum at which one day (or, if such amount due remains unpaid more than three Euro-Dollar Business Days, then for such other period of time as the Administrative Agent may elect which shall in no event be longer than six months) deposits in Dollars in an amount approximately equal to the amount of such overdue payment due to the Reference Bank are offered to the Reference Bank in

13

<Page>

the London interbank market for the applicable period determined as provided above by (ii) $1.00 \, \text{MINUS}$ the Euro-Dollar Reserve Percentage.

(d) The Administrative Agent shall determine each interest rate applicable to the Loans hereunder. The Administrative Agent shall give prompt notice to the Borrower and the Banks of each rate of interest so determined, and its determination thereof shall be conclusive in the absence of manifest error.

2.7 FEES.

- (a) FACILITY FEE. The Borrower shall pay to the Administrative Agent for the account of each Bank (i) a facility fee computed at the Facility Fee Rate (determined daily in accordance with the Pricing Schedule) on the total amount of such Bank's Commitment, regardless of usage; and (ii) for each day on which the outstanding principal amount of such Bank's loans exceeds 25% of such Bank's Commitment, a utilization fee computed at the Utilization Fee Rate (determined daily in accordance with the Pricing Schedule) on the outstanding principal amount of such Bank's Loans on such day. Such fees shall accrue from the date hereof through the Termination Date and shall be payable quarterly on the last day of each March, June, September and December.
- (b) The Borrower shall pay to the Administrative Agent for the account of each Bank on the date hereof an upfront fee equal to 0.05% of the amount of such Bank's Commitment on the date hereof. Such fee shall be fully earned when paid and shall be nonrefundable for any reason whatsoever.
- (c) The Borrower shall pay to the Administrative Agent for the account of each Bank on the six-month anniversary of the date hereof (or, if such day is not a Business Day, on the next succeeding Business Day), a participation fee equal to 0.05% of the amount of such Bank's Commitment on such date. Such fee shall be fully earned when paid and shall be nonrefundable for any reason whatsoever.
- 2.8 OPTIONAL TERMINATION OR REDUCTION OF COMMITMENTS. The Borrower shall have the right, upon at least five Domestic Business Days' prior written notice to the Administrative Agent, to terminate or reduce the unused portion of the Commitments. Any such reduction of the Commitments shall be in the minimum amount of \$10,000,000. The accrued facility fees with respect to the terminated Commitments shall be payable to the Administrative Agent for the account of the Banks on the effective date of such termination.
- 2.9 MANDATORY TERMINATION OR REDUCTION OF COMMITMENTS. (a) The Commitments shall terminate on the Termination Date, and any Loans then outstanding (together with accrued interest thereon) shall be due and payable

14

<Page>

(b) The Commitments shall be reduced automatically if the Borrower or any of its Subsidiaries shall at any time, or from time to time, after the date hereof receive any Net Cash Proceeds of any Reduction Event, by an amount equal to the largest integral multiple of \$1,000,000 which does not exceed such Net Cash Proceeds. The reductions in the Commitment required by this subsection shall be effective on the fifth Euro-Dollar Business Day following receipt by the Borrower or any of its Subsidiaries, as the case may be, of such Net Cash Proceeds; PROVIDED that if and to the extent such reduction would otherwise reduce the aggregate amount of the Commitments to an amount less than the related Dedicated Amount, such reduction shall, unless the Administrative Agent otherwise elects and so notifies the Borrower, be deferred so as to become effective simultaneously with reduction in the Dedicated Amount. For purposes of this section:

"Dedicated Amount" means the sum of the aggregate principal amount of Euro-Dollar Loans and the aggregate face amount of Supported Commercial Paper which, in each case, is outstanding at the time the Borrower or a Subsidiary receives Net Cash Proceeds of the related Reduction Event. The Dedicated Amount shall be reduced at each subsequent maturity of such Euro-Dollar Loans and such Supported Commercial Paper by the amount then maturing.

"Supported Commercial Paper" means commercial paper of the Borrower which requires liquidity support in the form of undrawn bank commitments and for which no such commitments other than the Commitments are available. At any time at which the Borrower has other committed bank facilities available as liquidity support for commercial paper, outstanding commercial paper shall be allocated first to such other facilities, so that only the amount which cannot be supported thereby shall constitute Supported Commercial Paper. Such allocation to other facilities shall be in inverse order of maturity, so that the earliest maturing commercial paper shall be Supported Commercial Paper.

The Borrower shall notify the Administrative Agent within two Euro-Dollar Business Days of receipt by it or a Subsidiary of Net Cash Proceeds of a Reduction Event, specifying the date and amount thereof and, if the provisions relating to Supported Commercial Paper are applicable, setting forth sufficient information with respect thereto to determine the resultant schedule for reduction of the Commitments.

2.10 VOLUNTARY PREPAYMENTS.

(a) The Borrower may, upon at least one Domestic Business Day's notice to the Administrative Agent, prepay Base Rate Loans without premium or penalty in whole at any time or from time to time in part in amounts aggregating \$10,000,000 or a higher integral multiple of \$1,000,000 by paying the principal amount being prepaid together with accrued interest thereon to the date of prepayment.

15

<Page>

- (b) Except as provided in Section 4.2, the Borrower may not prepay all or any portion of the principal amount of any Fixed Rate Loan prior to the maturity thereof.
- (c) Upon receipt of a notice of repayment pursuant to this Section 2.10, the Administrative Agent shall promptly notify each Bank of the contents thereof and of such Bank's ratable share of such repayment and such notice shall not thereafter be revocable by the Borrower.
- 2.11 MANDATORY PREPAYMENTS. On the date of any reduction of the Commitments pursuant to Section 2.9, the Borrower shall prepay such principal amount (together with accrued interest thereon) of, FIRST, outstanding Base Rate Loans, if any, and SECOND, outstanding Fixed Rate Loans, if any, as may be necessary so that after such repayment the aggregate outstanding principal amount of the Loans does not exceed the amount of the Commitments as then

reduced. Within the foregoing limits of this Section 2.11, each required payment or prepayment shall be made with respect to such outstanding Loans as the Borrower may designate to the Administrative Agent not less than three Business Days prior to the date required for such payment or prepayment or, failing such designation by the Borrower, as the Administrative Agent may specify by notice to the Borrower.

- GENERAL PROVISIONS AS TO PAYMENTS. The Borrower shall make each payment of principal of, and interest on, the Loans and of fees hereunder not later than 11:00 a.m. (New York City time) on the date when due in funds immediately available in New York City to the Administrative Agent at its address set forth on the signature pages hereof or at such other address as it may hereafter designate by notice to the Borrower and the Banks for the account of (i) the Domestic Lending Office of each Bank in the case of Domestic Loans or (ii) the Euro-Dollar Lending Office of each Bank in the case of Euro-Dollar Loans. The Administrative Agent will promptly distribute to each Bank its ratable share of each such payment received for the account of such Bank. Whenever any payment of principal of, or interest on, the Domestic Loans or of any fee shall be due on a day which is not a Domestic Business Day, the date for payment thereof shall be extended to the next succeeding Domestic Business Day. Whenever any payment of principal of, or interest on, the Euro-Dollar Loans shall be due on a day which is not a Euro-Dollar Business Day, the date for payment thereof shall be extended to the next succeeding Euro-Dollar Business Day unless as a result thereof it would fall in the next calendar month, in which case it shall be advanced to the next preceding Euro-Dollar Business Day. If the date for any payment of principal is extended by operation of law or otherwise, interest shall be payable for such extended time.
- 2.13 COMPUTATION OF INTEREST AND FEES. Interest on Loans made at the Prime Rate shall be computed on the basis of a year of 365 or 366 days, as the case may be, and paid for actual days elapsed. All other interest and fees shall be computed on the basis of a year of 360 days and paid for actual days elapsed.

16

<Page>

2.14 FUNDING LOSSES. If the Borrower makes any payment of principal with respect to any Fixed Rate Loan (pursuant to Section 2.11, Section 4 or Section 7 or otherwise) on any day other than the last day of an Interest Period applicable to such Loan, or the end of an applicable period fixed pursuant to the last paragraph of Section 2.6(c), or if the Borrower fails to borrow any Fixed Rate Loan after notice has been given to the Administrative Agent in accordance with Section 2.3, the Borrower shall reimburse each Bank on demand for any resulting loss or expense incurred by it including any loss incurred in obtaining, liquidating or employing deposits from third parties; PROVIDED that such Bank shall have delivered to the Borrower a certificate as to the amount of such loss, which certificate shall be conclusive in the absence of manifest error.

SECTION 3. CONDITIONS OF LENDING.

The obligation of the Banks to make each Loan hereunder is subject to the performance by the Borrower of all its obligations under this Agreement and to the satisfaction of the following further conditions:

- 3.1 ALL LOANS. In the case of each Loan hereunder (except for Loans made pursuant to Section 4), including the initial Loans:
 - (a) receipt by the Administrative Agent of the Notice of Borrowing as required by Section 2.3;
 - (b) the fact that immediately before and after the making of the Loan no Default or Event of Default shall have occurred and be continuing;
 - (c) the fact that the representations and warranties contained in this Agreement (except in the case of a Refunding Loan, the representation and warranty set forth in Section 5.4 as to any material adverse change which has theretofore been disclosed in writing by the Borrower to the Banks) are true on and as of the date of the Loan with the same force and effect as if made on and as of such date; and

(d) receipt by the Administrative Agent or the Banks of such other documents, evidence, materials and information with respect to the matters contemplated hereby as the Administrative Agent or the Banks may reasonably request.

Each Notice of Borrowing and borrowing by the Borrower hereunder shall be deemed to be a representation and warranty by the Borrower on the date of such borrowing as to the facts specified in (b) and (c) above.

3.2 INITIAL LOANS. In the case of the initial Loans:

17

<Page>

- (a) receipt by the Administrative Agent for the account of each Bank of a duly executed Note for such Bank;
- (b) receipt by the Administrative Agent of an opinion of Scott W. Johnson, Senior Vice President, General Counsel and Secretary of the Borrower dated the date of such Loans and substantially in the form of Exhibit B hereto;
- (c) receipt by the Administrative Agent of an opinion of Mayer, Brown & Platt, special counsel to the Administrative Agent, substantially in the form of Exhibit C hereto;
- (d) receipt by the Administrative Agent of certified copies of all corporate action taken by the Borrower to authorize the execution, delivery and performance of this Agreement and the Notes, and such other corporate documents and other papers as the Administrative Agent may reasonably request;
- (e) receipt by the Administrative Agent of a certificate of a duly authorized officer of the Borrower as to the incumbency, and setting forth a specimen signature, of each of the persons (i) who has signed this Agreement on behalf of the Borrower; (ii) who will sign the Notes on behalf of the Borrower; and (iii) who will, until replaced by other persons duly authorized for that purpose, act as the representatives of the Borrower for the purpose of signing documents in connection with this Agreement and the transactions contemplated hereby; and
- (f) receipt by the Administrative Agent of a certificate of a duly authorized officer of the Borrower to the effect set forth in Sections 3.1(b) and 3.1(c).

SECTION 4. CHANGE IN CIRCUMSTANCES AFFECTING FIXED RATE LOAMS.

4.1 BASIS FOR DETERMINING INTEREST RATE INADEQUATE. If with respect to any Interest Period (i) the Administrative Agent determines that deposits in Dollars (in the applicable amounts) are not being offered to the Reference Bank in the relevant market for such Interest Period, or (ii) Banks holding Notes evidencing at least 50% in aggregate principal amount of the Fixed Rate Loans (or the Commitments, if no Fixed Rate Loans are then outstanding) advise the Administrative Agent that the London Interbank Offered Rate or the CD Base Rate, as the case may be, as determined by the Administrative Agent will not adequately and fairly reflect the cost to such Banks of maintaining or funding their Euro-Dollar Loans or CD Loans, as the case may be, for such Interest Period, the Administrative Agent shall forthwith give notice thereof to the Borrower, whereupon the obligations of the Banks to make CD Loans or Euro-Dollar Loans, as the case may be, shall be suspended until the Administrative Agent notifies the Borrower that the circumstances giving rise to such suspension no longer exist. Unless the Borrower notifies the Administrative Agent at least two Domestic Business Days before the date of any Fixed Rate Loan for which a Notice

18

<Page>

Borrowing has previously been given that it elects not to borrow on such date, such Loans shall instead by made as Base Rate Loans.

- 4.2 ILLEGALITY. If, after the date of this Agreement, the adoption of any applicable law, rule or regulation, or any change therein, or any change in the interpretation or administration thereof by any governmental authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any Bank (or its Euro-Dollar Lending Office) with any request or directive (whether or not having the force of law) of any such authority, central bank or comparable agency shall make it unlawful or impossible for any Bank (or its Euro-Dollar Lending Office) to make, maintain or fund its Euro-Dollar Loans and such Bank shall so notify the Administrative Agent, the Administrative Agent shall forthwith so notify the other Banks and the Borrower, whereupon such Bank's obligation to make Euro-Dollar Loans shall be suspended until such Bank notifies the Administrative Agent and the Administrative Agent notifies the Borrower that the circumstances giving rise to such suspension no longer exist. Before giving any notice to the Administrative Agent pursuant to this Section 4.2, such Bank will designate a different Euro-Dollar Lending office if such designation will avoid the need for giving such notice and will not, in the sole judgment of such Bank, be otherwise disadvantageous to such Bank. If such Bank shall determine that it may not lawfully continue to maintain and fund any of its outstanding Euro-Dollar Loans to maturity and shall so specify in such notice, the Borrower shall immediately prepay in full the then outstanding principal amount of each such Euro-Dollar Loan, together with accrued interest thereon. Unless the Borrower notifies such Bank and the Administrative Agent to the contrary within two Euro-Dollar Business Days after receiving a notice from the Administrative Agent pursuant to this Section, the Borrower shall, concurrently with prepaying each such Euro-Dollar Loan, borrow a Base Rate Loan in an equal principal amount for an Interest Period coincident with the remaining term of the Interest Period applicable to such Euro-Dollar Loan.
- 4.3 INCREASED COSTS AND RATE OF RETURN. (a) If on or after the date hereof, in the case of any Loan or any obligation to make Loans, the adoption of any applicable law, rule or regulation, or any change in any applicable law, rule or regulation, or any change in the interpretation or administration thereof by any governmental authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any Bank (or its Euro-Dollar Lending Office) with any request or directive (whether or not having the force of law) of any such authority, central bank or comparable agency, shall (i) subject such Bank (or its Euro-Dollar Lending Office) to any tax, duty or other charge with respect to its Fixed Rate Loans, its Notes or its obligation to make Fixed Rate Loans, or shall change the basis of taxation of payments to such Bank (or its Euro-Dollar Lending Office) of the principal of or interest on its Fixed Rate Loans or in respect of any other amounts due under this Agreement in respect of its Fixed Rate Loans or its obligation to make Fixed Rate Loans (except for changes in the rate of tax on the overall net income of such Bank or its Euro-Dollar Lending Office imposed by the jurisdiction in which such Bank's principal executive office or Euro-Dollar Lending Office is located or (ii) impose, modify or deem applicable any reserve (including any such requirement imposed by the

1

<Page>

Board of Governors of the Federal Reserve System, but excluding (y) with respect to any CD Loan any such requirement included in an applicable CD Reserve Percentage and (z) with respect to any Euro-Dollar Loan any such requirement with respect to which such Bank is entitled to compensation during the relevant Interest Period under Section 2.14), special deposit, insurance assessment (excluding, with respect to any CD Loan, any such requirement reflected in an applicable Assessment Rate) or similar requirement against assets of, deposits with or for the account of, or credit extended by, such Bank's Euro-Dollar Lending Office or shall impose on such Bank (or its Euro-Dollar Lending Office) or on the United States market for certificates of deposit or the London interbank market any other condition affecting its Fixed Rate Loans, its Notes or its obligation to make Fixed Rate Loans, and the result of any of the foregoing is to increase the cost to such Bank (or its Euro-Dollar Lending Office) of making or maintaining any Fixed Rate Loan, or to reduce the amount of any sum received or receivable by such Bank (or its Euro-Dollar Lending Office) under this Agreement or under its Note with respect thereto, by an amount deemed by such Bank to be material, then, within 15 days after demand by such Bank (with a copy to the Administrative Agent), the Borrower shall pay to such Bank such additional amount or amounts as will compensate such Bank for such increased cost or reduction.

- (b) If any Bank shall have determined that, after the date hereof, the adoption of any applicable law, rule or regulation regarding capital adequacy, or any change in any such law, rule or regulation, or any change in the interpretation or administration thereof by any governmental authority, central bank or comparable agency charged with the interpretation or administration thereof, or any request or directive regarding capital adequacy (whether or not having the force of law) of any such authority, central bank or comparable agency, has or would have the effect of reducing the rate of return on capital of such Bank (or its parent) as a consequence of such Bank's obligations hereunder to a level below that which such Bank (or its parent) could have achieved but for such adoption, change, request or directive (taking into consideration such Bank's policies with respect to capital adequacy) by an amount deemed by such Bank to be material, then from time to time, within 15 days after demand by such Bank (with a copy to the Administrative Agent), the Borrower shall pay to such Bank such additional amount or amounts as will compensate such Bank (or its parent) for such reduction.
- (c) Each Bank will promptly notify the Borrower and the Administrative Agent of any event of which it has knowledge, occurring after the date hereof, which will entitle such Bank to compensation pursuant to this Section and will designate a different applicable lending office if such designation will avoid the need for, or reduce the amount of, such compensation and will not, in the judgment of such Bank, be otherwise disadvantageous to it. A certificate of any Bank claiming compensation under this Section and setting forth the additional amount or amounts to be paid to it hereunder shall be conclusive in the absence of manifest error. In determining such amount, such Bank may use any reasonable averaging and attribution methods.

20

<Page>

SECTION 5. REPRESENTATIONS AND WARRANTIES.

The Borrower hereby represents and warrants to the Banks that:

- 5.1 CORPORATE EXISTENCE AND POWER. The Borrower and each Subsidiary is a corporation duly organized and validly existing, and the Borrower and each Material Subsidiary is in good standing, under the laws of the State of its incorporation, has all power and authority to carry on its business as now being conducted and to own its properties and is duly licensed or qualified and in good standing as a foreign corporation in each other jurisdiction in which its properties are located or in which failure to qualify would materially and adversely affect the conduct of its business or the enforceability of contractual rights of the Borrower.
- 5.2 CORPORATE AUTHORIZATION. The execution, delivery and performance by the Borrower of this Agreement and the Notes are within the Borrower's corporate power, have been duly authorized by all necessary corporate action and will not contravene, or constitute a default under, any provision of applicable law or regulation or of the certificate of incorporation or by-laws of the Borrower, or of any judgment, order, decree, agreement or instrument binding on the Borrower or result in the creation of any Lien upon any of its property or assets.
- 5.3 BINDING EFFECT. This Agreement constitutes, and the Notes when duly executed on behalf of the Borrower and delivered in accordance with this Agreement will constitute, the valid and binding obligations of the Borrower, enforceable against the Borrower in accordance with their respective terms.

5.4 FINANCIAL STATEMENTS.

(a) The consolidated balance sheet of the Borrower and its Consolidated Subsidiaries as at December 31, 1999 and the related consolidated statements of income and cash flows of the Borrower and its Consolidated Subsidiaries for the fiscal year then ended, certified by PriceWaterhouseCoopers, LLP, certified public accountants, and set forth in the Borrower's 1999 Form 10-K, a copy of which has been delivered to each of the Banks, fairly present, in conformity with generally accepted accounting principles, the consolidated financial position of the Borrower and its Consolidated Subsidiaries at such date and the consolidated results of operations for such fiscal year.

(b) The unaudited consolidated balance sheet of the Borrower and its Consolidated Subsidiaries as at September 30, 2000 and the related consolidated statements of income and cash flows of the Borrower and its Consolidated Subsidiaries for the three months then ended, set forth in the Borrower's quarterly report for the fiscal quarter ended September 30, 2000 as filed with the Securities and Exchange Commission on Form 10-Q, a copy of which has been delivered to

21

<Page>

each of the Banks, fairly present in accordance with generally accepted accounting principles, the consolidated financial position of the Borrower and its Consolidated Subsidiaries as at such date and the consolidated results of operations for such period.

- (c) No material adverse change has occurred in the financial position, results of operations or business of the Borrower and its Consolidated Subsidiaries since December 31, 1999.
- 5.5 LITIGATION. There are no actions, suits or proceedings pending against or, to the knowledge of the Borrower, threatened against or affecting the Borrower or any Subsidiary in any court or before or by any governmental department, agency or instrumentality, an adverse decision in which could materially and adversely affect the financial condition or business of the Borrower or the ability of the Borrower to perform its obligations under this Agreement or the Notes.
- 5.6 TAXES. The Borrower has filed (or has obtained extensions of the time by which it is required to file) all United States federal income tax returns and all other material tax returns required to be filed by it and has paid all taxes shown due on the returns so filed as well as all other taxes, assessments and governmental charges which have become due, except such taxes, if any, as are being contested in good faith and as to which adequate reserves have been provided.
- 5.7 GOVERNMENTAL AND OTHER APPROVALS. No approval, consent or authorization of or filing or registration with any governmental authority or body is necessary for the execution, delivery or performance by the Borrower of this Agreement or the Notes or for the performance by the Borrower of any of the terms or conditions hereof or thereof, except for such approvals, consents or authorizations (copies of which have been delivered to the Banks) as have been obtained and are in full force and effect.
- 5.8 COMPLIANCE WITH ERISA. Each member of the Controlled Group has fulfilled its obligations under the minimum funding standards of ERISA and the Code with respect to each Plan and is in compliance in all material respects with the presently applicable provisions of ERISA and the Code, and has not incurred liabilities which are due and payable aggregating in excess of \$5,000,000 to the PBGC or a Plan under Title IV of ERISA.
- 5.9 ENVIRONMENTAL MATTERS. In the ordinary course of its business, the Borrower conducts an ongoing review of the effect of Environmental Laws on the business, operations and properties of the Borrower and its Subsidiaries, in the course of which it identifies and evaluates associated liabilities and costs (including any capital or operating expenditures required for clean-up or closure of properties presently or previously owned, any capital or operating expenditures required to achieve or maintain compliance with environmental protection standards imposed by law or as a condition of any license, permit or contract, any related constraints on operating activities, including

22

<Page>

any periodic or permanent shutdown of any facility or reduction in the level of or change in the nature of operations conducted thereat, any costs or liabilities in connection with off-site disposal of wastes or Hazardous Substances, and any actual or potential liabilities to third parties, including employees, and any related costs and expenses). On the basis of such review, the Borrower has reasonably concluded that such associated liabilities and costs, including the costs of compliance with Environmental Laws, are unlikely to have a material adverse effect on the financial

condition, business or results of operations of the Borrower and its Consolidated Subsidiaries, taken as a whole.

SECTION 6. COVENANTS.

So long as the Commitments shall be in effect or any Notes are outstanding, the Borrower agrees that:

- 6.1 FINANCIAL STATEMENTS. The Borrower will deliver to each of the Banks:
 - (a) as soon as available and in any event within 120 days after the end of each fiscal year of the Borrower, a consolidated balance sheet of the Borrower and its Consolidated Subsidiaries as at the end of such year, and consolidated statements of income and cash flows of the Borrower and its Consolidated Subsidiaries for such year, setting forth in each case in comparative form corresponding consolidated figures from the preceding fiscal year, all reported on in a manner acceptable to the Securities and Exchange Commission by PriceWaterhouseCoopers, LLP or other independent certified public accountants of nationally recognized standing;
 - (b) as soon as available and in any event within 45 days after the end of each of the first three quarters of each fiscal year of the Borrower, a consolidated balance sheet of the Borrower and its Consolidated Subsidiaries as at the end of such quarter and the related consolidated statements of income and cash flow of the Borrower and its Consolidated Subsidiaries for such quarter and for the portion of the Borrower's fiscal year ended at the end of such quarter setting forth in each case in comparative form the figures for the corresponding quarter and the corresponding portion of the Borrower's previous fiscal year, all certified (subject to normal year-end adjustments) as to fairness of presentation, generally accepted accounting principles and consistency by the chief financial officer or the chief accounting officer of the Borrower;
 - (c) simultaneously with the delivery of each set of financial statements referred to in clauses (a) and (b) above, a certificate of the chief financial officer or the chief accounting officer of the Borrower (i) setting forth in reasonable detail the calculations required to establish whether the Borrower was in compliance with the requirements of Sections 6.9 and 6.10 on the date of such financial statements and (ii) stating whether there exists on the date of such certificate any Default or Event of Default and, if any Default or Event of Default exists, setting forth the

23

<Page>

details thereof and the action which the Borrower is taking or proposes to take with respect hereto;

- (d) simultaneously with the delivery of each set of financial statements referred to in clause (a) above, a statement of the firm of independent public accountants which reported on such statements (i) to the effect that nothing has come to their attention to cause them to believe that there existed on the date of such statements any Default or Event of Default and (ii) confirming the calculations set forth in the officer's certificate delivered simultaneously therewith pursuant to clause (c) above;
- (e) forthwith upon the occurrence of any Default or Event of Default, a certificate of the chief financial officer or the chief accounting officer of the Borrower setting forth the details thereof and the action which the Borrower is taking or proposes to take with respect thereto;
- (f) promptly upon the mailing thereof to the shareholders of the Borrower generally, copies of all financial statements, reports and proxy statements so mailed;
 - (g) promptly upon the filing thereof, copies of all

registration statements (other than the exhibits thereto and any registration statements on Form S-8 or its equivalent) and annual, quarterly or monthly reports which the Borrower shall have filed with the Securities and Exchange Commission;

- (h) if and when any member of the Controlled Group (i) receives notice of complete or partial withdrawal liability or liabilities aggregating in excess of \$5,000,000 under Title IV of ERISA, a copy of such notice; or (ii) receives notice from the PBGC under Title IV of ERISA of an intent to terminate or appoint a trustee to administer any Plan or Plans having aggregate Unfunded Vested Liabilities in excess of \$5,000,000, a copy of such notice;
- (i) if at any time the value of all "margin stock" (as defined in Regulation U) owned by the Borrower and its Consolidated Subsidiaries exceeds (or would, following application of the proceeds of an intended Loan hereunder, exceed) 25% of the value of the total assets of the Borrower and its Consolidated Subsidiaries, in each case as reasonably determined by the Borrower, prompt notice of such fact and, promptly upon the request of any Bank, a duly completed statement of purpose on Form U-1 for each Bank together with such other information or documents as each Bank may be required to obtain under said Regulation U in connection with this Agreement; and
- (j) from time to time such additional information regarding the financial position or business of the Borrower as the Administrative Agent at the request of any Bank may reasonably request.

24

<Page>

- 6.2 MAINTENANCE OF EXISTENCE. Except as permitted by Section 6.12, the Borrower will, and will cause each Subsidiary to, preserve and maintain its corporate existence and all of its rights, privileges and franchises necessary or desirable in the normal conduct of its business, and will conduct its business in a regular manner.
- 6.3 MAINTENANCE OF PROPERTIES. The Borrower will, and will cause each Subsidiary to, keep all of its properties necessary, in the judgment of the Board of Directors of the Borrower, in its business in good working order and condition, ordinary wear and tear excepted, and will permit representatives of the Banks to inspect such properties, and to examine and make extracts from the books and records of the Borrower or any Subsidiary, during normal business hours.
- 6.4 COMPLIANCE WITH LAWS. The Borrower will, and will cause each Subsidiary to, comply with the requirements of all applicable laws, rules, regulations and orders of any governmental body or regulatory agency having jurisdiction, a breach of which could have a material adverse effect on the consolidated financial condition or the business taken as a whole of the Borrower and its Subsidiaries, except where contested in good faith and by proper proceedings.
- 6.5 NOTICE OF PROCEEDINGS. The Borrower will promptly give notice in writing to each Bank of all litigation, arbitral proceedings and regulatory proceedings affecting the Borrower or any Subsidiary or the property of the Borrower or any Subsidiary, except litigation or proceedings which, if adversely determined, could not materially and adversely affect the consolidated financial condition or the business taken as a whole of the Borrower and its Subsidiaries.
- 6.6 USE OF PROCEEDS. No part of the proceeds of any Loan hereunder will be used to purchase or carry any margin stock or to extend credit to others for the purpose of purchasing or carrying any margin stock. If requested by any Bank, the Borrower will furnish to any Bank in connection with any Loan hereunder a statement in conformity with the requirements of Federal Reserve Form U-1 referred to in Regulation U.
- 6.7 PAYMENT OF TAXES. The Borrower will, and will cause each Subsidiary to, pay and discharge all taxes, assessments and governmental charges or levies imposed on it or on its income or profits or on any of its property prior to the date on which penalties attach thereto, except that the Borrower or any Subsidiary will not be required hereby to pay any such tax, assessment, charge or levy the payment of which is being contested in good faith and by proper

proceedings and against which it is maintaining adequate reserves.

6.8 INSURANCE. The Borrower will, and will cause each Subsidiary to, maintain insurance with responsible companies in such amounts and against such risks as is usually carried by owners of similar businesses and properties in the same general areas in which the Borrower and its Subsidiaries operate.

25

<Page>

- 6.9 MAXIMUM CONSOLIDATED DEBT TO TOTAL CAPITAL RATIO. The Borrower will not permit the ratio of Consolidated Debt to Total Capital (expressed as a percentage) at any time to exceed 55%.
- 6.10 MINIMUM CONSOLIDATED NET WORTH. The Borrower will not permit Consolidated Net Worth at any time to be less than the sum of (i) \$648,758,000 PLUS (ii) 50% of the consolidated net income of the Borrower and its Consolidated Subsidiaries in each completed fiscal quarter of the Borrower ending after September 30, 2000 (with no deduction for a net loss in any such fiscal quarter).
- 6.11 NEGATIVE PLEDGE. Neither the Borrower nor any Subsidiary will create, assume or suffer to exist any Lien securing Debt on any asset now owned or hereafter acquired by it, except for:
 - (a) Liens existing on the date hereof securing Debt outstanding on the date hereof;
 - (b) any Lien existing on any asset of any corporation at the time such corporation becomes a Subsidiary and not created in contemplation of such event;
 - (c) any Lien on any asset securing Debt incurred or assumed for the purpose of financing all or any part of the cost of acquiring such asset; PROVIDED that such Lien attaches to such asset concurrently with or within 90 days after the acquisition thereof;
 - (d) any Lien on any asset of any corporation existing at the time such corporation is merged into or consolidated with the Borrower or a Subsidiary and not created in contemplation of such event;
 - (e) any Lien existing on any asset prior to the acquisition thereof by the Borrower or a Subsidiary and not created in contemplation of such acquisition;
 - (f) any Lien arising out of the refinancing, extension, renewal or refunding of any Debt secured by any Lien permitted by any of the foregoing clauses of this Section; PROVIDED that such Debt is not increased and is not secured by any additional assets;
 - (g) any Lien arising pursuant to any order of attachment, distraint or similar legal process arising in connection with court proceedings so long as the execution or other enforcement thereof is effectively stayed and the claims secured thereby are being contested in good faith by appropriate proceedings; and
 - (h) Liens not otherwise permitted by the foregoing clauses of this Section securing Debt in aggregate principal amount not to exceed 4% of the

26

<Page>

consolidated assets of the Borrower and the Consolidated Subsidiaries at any time outstanding.

6.12 CONSOLIDATIONS, MERGERS AND SALES OF ASSETS. The Borrower will not (i) consolidate or merge with or into any other Person unless the Borrower shall be the surviving corporation or (ii) sell, lease or otherwise transfer (whether in one transaction or in a series of transactions) all or any substantial part

of its assets to any other Person. The Borrower will not permit any Subsidiary to consolidate with, merge with or into or sell, lease or otherwise transfer (whether in one transaction or in a series of transactions) all or any substantial part of its assets to any Person other than the Borrower or a Wholly-Owned Consolidated Subsidiary.

For purposes of this Section 6.12, "substantial part" means 15% or more of the consolidated assets of the Borrower and the Consolidated Subsidiaries.

SECTION 7. EVENTS OF DEFAULT.

If any one or more of the following events ("Events of Default") shall have occurred and be continuing:

- (a) the Borrower shall fail to pay any principal of any Note when $\mbox{due;}$ or
- (b) the Borrower shall fail to pay any interest on any Note, any fee or any other amount due hereunder or under the Notes when due and such failure shall continue for five consecutive days; or
- (c) the Borrower shall fail to perform or observe any of the covenants contained in Section 6.1(e) or Sections 6.9 to 6.12 (inclusive); or
- (d) any representation and warranty made by the Borrower herein or in any instrument or document delivered pursuant hereto shall prove to be incorrect or misleading in any material respect upon the date when made; or
- (e) the Borrower shall fail to perform any term, covenant or agreement contained herein (other than those specified in clauses (a), (b) or (c) above) for 30 days after written notice thereof has been given to the Borrower by the Administrative Agent at the request of any Bank; or
- (f) the Borrower or any Subsidiary shall (i) fail to pay any Debt (other than the Notes) when due or interest thereon and such failure shall continue for more than any applicable period of grace with respect thereto, or (ii) fail to observe or perform any term, covenant or agreement contained in any agreement or instrument (other than this Agreement or the Notes) by which it is bound

27

<Page>

evidencing or securing or relating to any Debt, if the effect thereof is to permit (or, with the giving of notice or lapse of time or both, would permit) the holder or holders thereof or of any obligations issued thereunder or a trustee or trustees acting on behalf of such holder or holders to cause acceleration of the maturity thereof or of any such obligation; PROVIDED, that the aggregate amount of Debt with respect to which any such event or condition shall have occurred shall equal or exceed \$1,000,000; or

- (g) the Borrower or any Material Subsidiary shall commence a voluntary case or other proceeding seeking liquidation, reorganization or other relief with respect to itself or its debts under any bankruptcy, insolvency or other similar law now or hereafter in effect or seeking the appointment of a trustee, receiver, liquidator, custodian or other similar official of it or any substantial part of its property, or shall consent to any such relief or to the appointment of or taking possession by any such official in an involuntary case or other proceeding commenced against it, or shall make a general assignment for the benefit of creditors, or shall fail generally to pay its debts as they become due, or shall take any corporate action to authorize any of the foregoing; or
- (h) an involuntary case or other proceeding shall be commenced against the Borrower or any Material Subsidiary seeking liquidation, reorganization or other relief with respect to it or its debts under any bankruptcy, insolvency or other similar law now or hereafter in effect or seeking the appointment of a trustee, receiver, liquidator,

custodian or other similar official of it or any substantial part of its property, and such involuntary case or other proceeding shall remain undismissed and unstayed for a period of 60 days; or an order for relief shall be entered against the Borrower or any Material Subsidiary under the federal bankruptcy laws as now or hereafter in effect; or

(i) the Borrower or any other member of the Controlled Group shall fail to pay when due any amount or amounts aggregating in excess of \$5,000,000 which it shall have become liable to pay to the PBGC or to a Plan under Title IV of ERISA; or notice of intent to terminate a Plan or Plans having aggregate Unfunded Vested Liabilities in excess of \$5,000,000 shall be filed under Title IV of ERISA by any member of the Controlled Group, any plan administrator or any combination of the foregoing; or the PBGC shall institute proceedings under Title IV of ERISA to terminate or to cause a trustee to be appointed to administer any Plan or Plans having aggregate Unfunded Vested Liabilities in excess of \$5,000,000 or a proceeding shall be instituted by a fiduciary of any Plan against any member of the Controlled Group to enforce Section 515 of ERISA with respect to any amount or amounts aggregating in excess of \$5,000,000 and such proceeding shall not have been dismissed within 30 days thereafter; or a condition shall exist by reason of which the PBGC would be entitled to obtain a decree adjudicating that any Plan or Plans having aggregated Unfunded Vested Liabilities in excess of \$5,000,000 must be terminated; or

28

<Page>

- (j) judgments or orders for the payment of money in excess of \$1,000,000 in the aggregate shall be rendered against the Borrower or any Subsidiary and such judgments or orders shall continue unsatisfied and unstayed for a period of 30 days; or
 - (k) any Change of Control shall occur;

then, and in every such event, (1) in the case of any of the Events of Default specified in paragraphs (g) or (h) above, the Commitments shall thereupon automatically be terminated and the principal of and accrued interest on the Notes shall automatically become due and payable without presentment, demand, protest or other notice or formality of any kind, all of which are hereby expressly waived and (2) in the case of any other Event of Default specified above, the Administrative Agent shall, if requested by the Required Banks, by notice in writing to the Borrower, terminate the Commitments hereunder, if still in existence, and they shall thereupon be terminated, and the Administrative Agent shall, if requested by the Required Banks, by notice in writing to the Borrower, declare the Notes and all other sums payable under this Agreement to be, and the same shall thereupon forthwith become, due and payable without presentment, demand, protest or other notice or formality of any kind, all of which are hereby expressly waived.

SECTION 8. THE ADMINISTRATIVE AGENT.

- 8.1 APPOINTMENT AND AUTHORIZATION. Each Bank irrevocably appoints and authorizes the Administrative Agent to take such action as agent on its behalf and to exercise such powers under this Agreement and the Notes as are delegated to the Administrative Agent by the terms hereof or thereof, together with all such powers as are reasonably incidental thereto.
- 8.2 ADMINISTRATIVE AGENT AND AFFILIATES. Morgan Guaranty Trust Company of New York shall have the same rights and powers under this Agreement as any other Bank and may exercise or refrain from exercising the same as though it were not the Administrative Agent, and Morgan Guaranty Trust Company of New York and its affiliates may accept deposits from, lend money to, and generally engage in any kind of business with the Borrower or any Subsidiary or Affiliate of the Borrower as if it were not the Administrative Agent hereunder.
- 8.3 ACTION BY ADMINISTRATIVE AGENT. The obligations of the Administrative Agent hereunder are only those expressly set forth herein. Without limiting the generality of the foregoing, the Administrative Agent

shall not be required to take any action with respect to any Default, except as expressly provided in Section 7.

8.4 CONSULTATION WITH EXPERTS. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent public accountants and other experts selected by it and shall not be liable for any action taken or omitted to be

29

<Page>

taken by it in good faith in accordance with the advice of such counsel, accountants or experts.

- 8.5 LIABILITY OF ADMINISTRATIVE AGENT. Neither the Administrative Agent nor any of its directors, officers, agents, or employees shall be liable for any action taken or not taken by it in connection herewith (i) with the consent or at the request of the Required Banks or (ii) in the absence of its own gross negligence or willful misconduct. Neither the Administrative Agent nor any of its directors, officers, agents or employees shall be responsible for or have any duty to ascertain, inquire into or verify (a) any statement, warranty or representation made in connection with this Agreement or any Borrowing hereunder; (b) the performance or observance of any of the covenants or agreements of the Borrower; (c) the satisfaction of any condition specified in Section 3, except receipt of items required to be delivered to the Administrative Agent; or (d) the validity, effectiveness or genuineness of this Agreement, the Notes or any other instrument or writing furnished in connection herewith. The Administrative Agent shall not incur any liability by acting in reliance upon any notice, consent, certificate, statement, or other writing (which may be a bank wire, telex, facsimile or similar writing) believed by it to be genuine or to be signed by the proper party or parties. As to any matters not expressly provided for by this Agreement, the Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, hereunder in accordance with instructions signed by the Required Banks, and such instructions of the Required Banks and any action taken or failure to act pursuant thereto shall be binding on all of the Banks.
- 8.6 INDEMNIFICATION. Each Bank shall, ratably in accordance with its Commitment, indemnify the Administrative Agent (to the extent not reimbursed by the Borrower) against any cost, expense (including counsel fees and disbursements), claim, demand, action, loss or liability (except such as result from the Administrative Agent's gross negligence or willful misconduct) that the Administrative Agent may suffer or incur in connection with this Agreement or any action taken or omitted by the Administrative Agent hereunder.
- 8.7 FAILURE TO ACT. Except for action expressly required of the Administrative Agent hereunder the Administrative Agent shall in all cases be fully justified in failing or refusing to act hereunder unless it shall be indemnified to its satisfaction by the Banks against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action.
- 8.8 RESIGNATION OR REMOVAL OF ADMINISTRATIVE AGENT. Subject to the appointment and acceptance of a successor Administrative Agent as provided below, the Administrative Agent may resign at any time by giving notice thereof to the Banks and the Borrower and the Administrative Agent may be removed at any time with or without cause by the Required Banks with the prior written consent of the Borrower. Upon any such resignation or removal, the Required Banks shall have the right to appoint a successor Administrative Agent. If no successor Administrative Agent shall have been so appointed by the Required Banks and shall have accepted such appointment within 30 days after the

30

Required Bank's removal of the retiring Administrative Agent, then the retiring Administrative Agent may, on behalf of the Banks, appoint a successor Administrative Agent, which shall be a bank which has an office in New York, New York. Upon the acceptance of any appointment as Administrative Agent hereunder by a successor Administrative Agent, such successor Administrative Agent shall thereupon succeed to and become vested with all rights, powers, privileges and duties of the retiring Administrative Agent, and the retiring Administrative Agent shall be discharged from its duties and obligations hereunder. After any retiring Administrative Agent's resignation or removal hereunder as Administrative Agent, the provisions of this Section 8 shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as the Administrative Agent.

8.9 CREDIT DECISION. Each Bank acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Bank, and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Bank also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Bank, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking any action under this Agreement.

SECTION 9. MISCELLANEOUS.

- 9.1 NOTICES. Unless otherwise specified herein all notices, requests, demands or other communications to or from the parties hereto shall be deemed to have been duly given and made when sent by United States mail, certified, return receipt requested, or by facsimile, when sent and receipt is electronically confirmed; PROVIDED that notices to the Administrative Agent pursuant to Sections 2.3 and 2.10 shall not be effective until received by the Administrative Agent. Any such notice, request, demand or communication shall be delivered or addressed as follows:
 - (a) if to any party hereto, to it at its address or facsimile number set forth on the signature pages hereof; and
 - (b) if to any holder of a Note, other than a Bank, to it at the address or facsimile number of the original payee thereof or at the address or facsimile number of any subsequent holder if notice of the transfer of such Note and the name and the address or facsimile number of such subsequent holder shall have been given to the Administrative Agent and the Borrower;

or at such other address or facsimile number as any party hereto or any subsequent holder may designate by written notice to the Administrative Agent and the Borrower.

9.2 AMENDMENTS AND WAIVERS; CUMULATIVE REMEDIES.

31

<Page>

- (a) None of the terms of this Agreement may be waived, altered or amended except by an instrument in writing duly executed by the Borrower and the Required Banks (and, if the rights or duties of the Administrative Agent are affected thereby, by the Administrative Agent); PROVIDED that no such amendment or waiver shall, unless signed by all the Banks, (i) increase the Commitment of any Bank or subject any Bank to any additional obligation, (ii) reduce the principal of or rate of interest on the Notes or any fees hereunder, (iii) postpone the date fixed for any payment of principal of or interest on the Notes or any fees hereunder or (iv) change the percentage of the Commitments or of the aggregate unpaid principal amount of the Notes, or the number of Banks, which shall be required for the holders of Notes or the Banks or any of them to take any action hereunder.
- (b) No failure or delay on the part of the Administrative Agent, any Bank, or the holder of any Note in exercising any right, power or privilege under this Agreement or the Notes shall operate as a

waiver thereof, nor shall any single or partial exercise of any right, power or privilege under this Agreement or the Notes preclude any other or further exercise thereof or the exercise of any other right, power or privilege. Tho rights and remedies provided in and contemplated by this Agreement and the Notes are cumulative and not exclusive of any rights or remedies provided by law.

9.3 SUCCESSORS AND ASSIGNS.

- (a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of each Bank (and any attempted assignment or transfer by the Borrower without such consent shall be null and void).
- (b) Any Bank may assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans at the time owing to it); PROVIDED that (i) except in the case of an assignment of the entire remaining amount of the assigning Bank's Commitment, the amount of the Commitment of the assigning Bank subject to each such assignment (determined as of the date the Assignment and Acceptance with respect to such assignment is delivered to the Administrative Agent) shall not be less than \$5,000,000 unless each of the Administrative Agent and, so long as an Event of Default has not occurred, the Borrower otherwise consent, (ii) each partial assignment shall be made as an assignment of a proportionate part of all of the assigning Bank's rights and obligations under this Agreement, and (iii) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Acceptance, together with a processing and recordation fee of \$3,500, and the Eligible Assignee, if it shall not be a Bank, shall deliver to the Administrative Agent an Administrative Questionnaire. Subject to acceptance and

32

<Page>

recording thereof pursuant to paragraph (d) of this Section, from and after the effective date specified in each Assignment and Acceptance, the Eliqible Assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Assignment and Acceptance, have the rights and obligations of a Bank under this Agreement, and the assigning Bank thereunder shall, to the extent of the interest assigned by such Assignment and Acceptance, be released from its obligations under this Agreement (and, in the case of an Assignment and Acceptance covering all of the assigning Bank's rights and obligations under this Agreement, such Bank shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 2.14, 4 and 9.4). If the Eligible Assignee is not incorporated under the laws of the United States of America or a state thereof, it shall deliver to the Borrower and the Administrative Agent certification as to exemption from deduction or withholding of any United States federal income taxes in connection herewith. Promptly upon the effectiveness of any assignment to an Eligible Assignee which was not previously a Bank, the Borrower shall execute a new Note payable to such Eligible Assignee. If any Bank ceases to be a party hereto as a result of an assignment, such Bank shall promptly deliver its Note to the Borrower for cancellation. Any assignment or transfer by a Bank of rights or obligations under this Agreement that does not comply with this paragraph shall be treated for purposes of this Agreement as a sale by such Bank of a participation in such rights and obligations in accordance with paragraph (d) of this Section.

(c) Upon its receipt of a duly completed Assignment and Acceptance executed by an assigning Bank and an Eligible Assignee, the Eligible Assignee's completed Administrative Questionnaire (unless the Eligible Assignee shall already be a Bank hereunder) and the processing and recordation fee referred to in paragraph (b) of this Section, the

Administrative Agent shall accept such Assignment and Acceptance and record the information contained therein in a register for the recordation of the names and addresses of the Banks, and the Commitments of, and principal amount of the Loans owing to, each Bank pursuant to the terms hereof from time to time (the "Register"). No assignment shall be effective for purposes of this Agreement unless it has been recorded in the Register as provided in this paragraph.

(d) Any Bank may, without the consent of the Borrower or the Administrative Agent, sell participations to one or more banks or other Persons (a "Participant") in all or a portion of such Bank's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans owing to it); PROVIDED that (i) such Bank's obligations under this Agreement shall remain unchanged, (ii) such Bank shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent and the other Banks shall continue to deal solely and directly with such Bank in connection with such Bank's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Bank sells such a participation shall provide that such Bank shall retain the sole right to enforce this Agreement and to

33

<Page>

approve any amendment, modification or waiver of any provision of this Agreement; PROVIDED that such agreement or instrument may provide that such Bank will not, without the consent of the Participant, agree to any amendment, modification or waiver described in the proviso to Section 9.2(a) that affects such Participant. Subject to paragraph (e) of this Section, each Participant shall be entitled to the benefits of Sections 4 and 2.14 to the same extent as if it were a Bank and had acquired its interest by assignment pursuant to paragraph (b) of this Section. To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 9.4 as though it were a Bank; PROVIDED that such Participant agrees to be subject to Section 9.5 as though it were a Bank.

- (e) A Participant shall not be entitled to receive any greater payment under Section 2.14 or Section 4 than the applicable Bank would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent.
- (f) Without limiting any right of a Bank to pledge its rights under this Agreement to a third party, it is specifically acknowledged that any Bank may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Bank, including any pledge or assignment to secure obligations to a Federal Reserve Bank, and this Section shall not apply to any such pledge or assignment of a security interest; PROVIDED that no such pledge or assignment of a security interest shall release a Bank from any of its obligations hereunder or substitute any such pledgee or assignee for such Bank as a party hereto.
- 9.4 INDEMNIFICATION BY THE BORROWER; DOCUMENTARY TAXES. (a) The Borrower agrees to indemnify the Administrative Agent and each Bank, their respective affiliates and the respective directors, officers, agents and employees of the foregoing (each an "Indemnified Person") and hold each Indemnified Person harmless from and against any and all liabilities, losses, damages, costs and expenses of any kind, including, without limitation, the reasonable fees and disbursements of counsel, which may be incurred by such Indemnified Person in connection with any investigative, administrative or judicial proceeding (whether or not such Indemnified Person shall be designated a party thereto) brought or threatened relating to or arising out of this Agreement or any Note or any actual or proposed use of proceeds of any Loan hereunder; provided that no Indemnified Person shall have the right to be indemnified hereunder for such Indemnified Person's own gross negligence or willful misconduct as determined by a court of competent jurisdiction.
- (b) The Borrower shall pay all out-of-pocket expenses of the Administrative Agent (including fees and disbursements of special counsel for the Banks) in connection with the preparation and administration of this

Agreement, the Notes and any waiver or amendment of any provision hereof or thereof and, if there is an Event of Default, all out-of-pocket expenses incurred by the Administrative Agent or any Bank (including fees and

3.4

<Page>

disbursements of counsel and time charges of attorneys who may be employees of the Administrative Agent or such Bank) in connection with such Event of Default and collection and other enforcement proceedings resulting therefrom. The Borrower agrees to indemnify the Banks from and hold them harmless against any documentary taxes, assessments or charges made by any governmental authority by reason of the execution and delivery of this Agreement or the Notes.

- 9.5 SHARING OF SET-OFFS. Each Bank agrees that if it shall, by exercising any right of set-off or counterclaim or otherwise, receive payment of a proportion of the aggregate amount of principal and interest due with respect to any Note held by it which is greater than the proportion received by any other Bank in respect of the aggregate amount of principal and interest due with respect to any Note held by such other Bank, the Bank receiving such proportionately greater payment shall purchase such participations in the Notes held by the other Banks, and such other adjustments shall be made, as may be required so that all such payments of principal and interest with respect to Notes held by Banks shall be shared by the Banks pro rata; PROVIDED that nothing in this Section 9.5 shall impair the right of any Bank to exercise any right of set-off or counterclaim it may have and to apply the amount subject to such exercise to the payment of indebtedness of the Borrower other than the indebtedness evidenced by the Notes. The Borrower agrees, to the fullest extent it may effectively do so under applicable law, that any holder of a participation in a Note, whether or not acquired pursuant to the foregoing arrangements, may exercise rights of set-off or counterclaim and other rights with respect to such participation as fully as if such holder of a participation were a direct creditor of the Borrower in the amount of such participation. If under any applicable bankruptcy, insolvency or other similar law, any Bank receives a secured claim in lieu of a set-off to which this Section would apply, such Bank shall, to the extent practicable, exercise its rights in respect of such secured claim in a manner consistent with the rights of the Banks entitled under this Section to share in the benefits of any recovery on such secured claim.
- 9.6 COLLATERAL. Each of the Banks represents that it in good faith is not relying on any "margin stock" (as defined in Regulation U) as collateral in the extension or maintenance of the credit provided for in this Agreement.
- 9.7 COUNTERPARTS. This Agreement may be signed in any number of counterparts with the same effect as if the signatures thereto and hereto were upon the same instrument.
- 9.8 HEADINGS; TABLE OF CONTENTS. The section and subsection headings used herein and the Table of Contents have been inserted for convenience of reference only and do not constitute matters to be considered in interpreting this Agreement.
- 9.9 GOVERNING LAW. This Agreement and the Notes shall be construed in accordance with and governed by the law of the State of New York.
- 9.10 DOCUMENTATION AGENT. No Bank identified on the facing page of this Agreement or otherwise herein, or in any amendment hereof or other document related

35

<Page>

hereto, as being the "Documentation Agent" shall have any right, power, obligation, liability, responsibility or duty under this Agreement in such capacity. Each Bank acknowledges that it has not relied, and will not rely, on any Person so identified in deciding to enter into this Agreement or in taking or refraining from taking any action hereunder or pursuant hereto.

IN WITNESS WHEREOF, the parties have caused this Agreement to be duly executed and delivered by their proper and duly authorized officers as of the day and year first above written.

BEMIS COMPANY, INC.

By: /s/ Benjamin R. Field, III

Title: Senior Vice President,
Chief Financial Officer
and Treasurer

MORGAN GUARANTY TRUST COMPANY OF NEW YORK, as Administrative Agent

By: /s/ Susan L. Pearson
----Title: Vice President

S-1

<Page>

COMMITMENTS \$65,000,000

BANK ONE, NA (Main Office Chicago), as Documentation Agent and as a Bank $\,$

By: /s/ [ILLEGIBLE]

Title: AVP

Domestic and Euro-Dollar Lending Office:

1 Bank One Plaza 14th Floor IL1-0173 Chicago, Illinois 60670 Attention: J. Garland Smith Facsimile: (312) 732-3888

S-2

<Page>

\$40,000,000

WACHOVIA BANK, N.A.

By: /s/ Shawn Janko

Title: AVP

Domestic and Euro-Dollar Lending Office:

191 Peachtree Street N.E. Suite 2800

Atlanta, GA 30303 Attention: Shawn Janko Facsimile: (404) 332-6898 with a copy to:

Wachovia Bank, N.A.
70 West Madison Street, Suite 2440
Chicago, Illinois 60602

Attention: John C. Canty Facsimile: (312) 853-0693

S-3

<Page>

\$20,000,000

WELLS FARGO BANK, NATIONAL ASSOCIATION

By: /s/ Molly S. Van Metre

Title: Vice President and Senior Banker

By: /s/ Chad M. Kortgard

Title: Banking Officer

Domestic and Euro-Dollar
 Lending Office:

U.S. Corporate Banking
MAC N9305-031
Sixth and Marquette
Minneapolis, Minnesota 55479
Attention: Molly S. Van Metre
Facsimile: (612) 667-2276

S-4

<Page>

\$40,000,000

U.S. BANK NATIONAL ASSOCIATION

By: /s/ Elliot J. Jaffe

Title: Senior Vice President

Domestic and Euro-Dollar
 Lending Office:

U.S. Bank Place MPFP0702

601 Second Avenue South

Minneapolis, Minnesota 55402-4302

Attention: Elliot J. Jaffee Facsimile: (612) 973-0825

S-5

<Page>

\$65,000,000

MORGAN GUARANTY TRUST COMPANY OF NEW YORK

By: /s/ Susan L. Pearson

Title: Vice President

Domestic and Euro-Dollar Lending Office:

60 Wall Street

New York, New York 10260-0060 Attention: Loan Capital Markets Facsimile: (212) 648-5336

S-6

<Page>

PRICING SCHEDULE

The "Euro-Dollar Margin", "CD Margin", "Facility Fee" and "Utilization Fee" for any day are the respective percentages set forth below in the applicable row under the column corresponding to the Status that exists on such day:

<Table> <Caption>

	Level I Status	Level II Status	Level III Status
	`	(At least BBB	(Neither Level I
	-	by S&P OR Baa2	Status nor Level II
	Moody's) *	by Moody's)	Status applies)
<\$>	<c></c>	<c></c>	<c></c>
Euro-Dollar Margin	0.400%	0.575%	0.800%
CD Margin	0.525%	0.700%	0.925%
Facility Fee	0.100%	0.125%	0.200%
Utilization Fee (>25%)		0.125%	0.250%
001112d010H fee (223%)	U.12J6	U.12J6	0.230%

 | | |*Ratings in effect at any date refer to the ratings of the Borrower's long-term unsecured non credit-enhanced debt in effect at the close of business on such date.

If the Borrower is split-rated and the ratings differential is one notch, the higher of the two ratings will apply (e.g., A-/Baal results in Level I Status and BBB/Baa3 results in Level II Status). If the Company is split-rated and the ratings differential is two or more notches, the rating which is one notch above the lower rating shall be used (e.g., A-/Baa3 results in Level II Status and BBB+/Bal results in Level III Status). If at any date, the Borrower's long-term unsecured debt is rated by neither S&P nor Moody's, then Level III shall apply.

PS-1

March 14, 2001

Bemis Company, Inc. 222 South 9th Street Suite 2300 Minneapolis, MN 55402-4099

Re: ADDITION OF BANK UNDER BRIDGE CREDIT AGREEMENT

Ladies/Gentlemen:

Please refer to the Bridge Credit Agreement dated as of January 12, 2001 among Bemis Company, Inc. (the "Company"), various financial institutions and Morgan Guaranty Trust Company of New York, as Administrative Agent (the "Credit Agreement"; capitalized terms used but not defined herein have the respective meanings set forth in the Credit Agreement).

The Company, the Banks and the Administrative Agent agree that, effective on the date on which the Administrative Agent has received (i) a counterpart of this letter agreement executed by the Company, the Banks and BBL International (U.K.) Limited (the "New Bank") and (ii) a Note payable to the New Bank, the New Bank will become a Bank under the Credit Agreement with a Commitment of \$20,000,000.

By its signature below, the New Bank (a) acknowledges that it has received a copy of the Credit Agreement and the Schedules and Exhibits thereto, together with such other documents and information as it has deemed appropriate to make its own credit and legal analysis and decision to become a Bank under the Credit Agreement, (b) agrees that it will, independently and without reliance upon the Administrative Agent or any other Bank and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit and legal decisions in taking or not taking action under the Credit Agreement and (c) agrees to be bound by the terms of the Credit Agreement as fully and to the same extent as if it were an original Bank under the Credit Agreement.

<Page>

This letter shall be governed by the laws of the State of New York applicable to contracts made and to be performed entirely within such State.

Very truly yours,

MORGAN GUARANTY TRUST COMPANY OF NEW YORK, as Administrative Agent and as a Bank

By:
Title:
BANK ONE, NA (Main Office Chicago), as Documentation Agent and as a Bank
By: Title:
WACHOVIA BANK, N.A.
By: Title:
WELLS FARGO BANK, NATIONAL ASSOCIATION
By: Title:
By: Title:
U.S. BANK NATIONAL ASSOCIATION
2
By: Title:
New Bank:
BBL INTERNATIONAL (U.K.) LIMITED
By:

<Page>

	Title:
	By: Title:
	Domestic and Euro-Dollar Lending Office:
	[Address] Attention: Facsimile:
Acknowledged and Agreed:	
BEMIS COMPANY, INC.	
By: Title:	

2000

Three Months Ended March 31,

2001

2000

COMPUTATION OF RATIO OF EARNINGS TO FIXED CHARGES BEMIS COMPANY, INC.

1996

For the Year Ended December 31,

1998

1999

1997

	(unau	dited)					(aı	ıdited)		
				(in thousan	ds of d	lollars except	ratio (lata)		
EARNINGS:										
Earnings before income taxes and cumulative effect of accounting changes	\$ 47,743	\$	48,187	\$ 166,037	\$	164,924	\$	165,030	\$ 185,875	\$ 211,502
Fixed charges	6,964		11,498	17,405		24,920		27,562	26,546	36,361
Less: Capitalized interest	(466)		(42)	(787)		(1,317)		(1,331)	(1,568)	(1,653)
Minority interest in net income of majority-owned subsidiaries	46		99	4,695		5,406		4,496	4,224	496
Less: Minority interest in net income of majority-owned subsidiaries without fixed charges	(51)		(103)	0		0		(242)	(185)	(384)
Losses recognized in pre- tax income of less than 50% owned persons	649		694	0		0		1,546	7,614	2,467
Capitalized interest amortization	192		159	577		509		801	769	760
TOTAL EARNINGS	55,077		60,492	187,927		194,442		197,862	223,275	249,549
FIXED CHARGES:										
Interest expense:										
Consolidated interest expense	5,723		10,681	13,397		18,893		21,866	21,218	31,609
Capitalized interest	466		42	787		1,317		1,331	1,568	1,653
Total interest expense	6,189		10,723	14,184		20,210		23,197	22,786	33,262
Interest inherent in rent expense	775		775	3,221		4,710		4,365	3,760	3,099
TOTAL FIXED CHARGES	6,964		11,498	17,405		24,920		27,562	26,546	36,361

QuickLinks

COMPUTATION OF RATIO OF EARNINGS TO FIXED CHARGES BEMIS COMPANY, INC.

Exhibit 99



BEMIS COMPANY, INC.

222 South Ninth Street Suite 2300 Minneapolis, MN 55402-4099

For additional information please contact:

Melanie E. R. Miller, Director of Investor Relations (612) 376-3030

FOR IMMEDIATE RELEASE

BEMIS COMPANY REPORTS RECORD SALES AND EARNINGS DURING SECOND OUARTER

MINNEAPOLIS, July 26, 2001—Bemis Company, Inc. (NYSE-BMS) reported record quarterly diluted earnings of \$0.67 per share, up from \$0.66 per share in the second quarter last year and in line with Company guidance of at or slightly above the prior year's level. Net sales rose for the ninth consecutive quarter to a record \$582 million, an 8% increase over last year's second quarter.

Flexible packaging, representing 79% of total company sales, reported net sales of \$458 million in the second quarter, an increase of 14% over the prior year. Excluding acquisitions, organic growth was about 3%. Operating profit for the second quarter was \$72 million, up 31% from the prior year's period. Manufacturing efficiencies resulting from new manufacturing lines installed during 2000 drove margin improvement. Paper products also delivered improved results as the benefits of the restructuring efforts of the late 1990's are being realized.

Sales from our pressure sensitive materials business segment were \$124 million, an 8% decrease from the strong performance of the prior year's quarter. This segment reported break-even operating results for the quarter, including charges totaling approximately \$2.6 million related to severance costs, the closure of two small, ancillary operations, and other non-recurring charges. These results compare to the prior year's profit of \$14.2 million, recorded before the economic downturn in the second half of 2000.

"Our flexible packaging business continues to deliver outstanding performance and benefit from last year's strategic acquisitions," said Jeff Curler, President and Chief Executive Officer of Bemis Company. "Late in the first half and during all of the second half of 2000, we were fighting capacity constraints in many parts of our flexible packaging business. Through acquisitions and capital additions implemented during the latter half of 2000, we have significantly improved manufacturing efficiencies and capacity. These acquisitions also brought benefits to our top line and are allowing us to bring innovative new products to the marketplace. Our high barrier and polyethylene product lines continue to maintain strong market positions in the businesses in which we choose to participate. I am also pleased to report that the paper product line has successfully solidified its business base and markedly improved performance.

The recently announced acquisition of Duralam will strengthen our position in key flexible packaging markets and provide capacity for future growth. In addition, we recently purchased the land, building and manufacturing equipment of the Prattville, Alabama facility of Wright Plastics that had been closed earlier this year. This modern plant will produce polyethylene packaging products. We intend to focus on several markets at this plant and it will complement our existing facilities.

Operating results of our pressure sensitive materials segment continue to be disappointing, and we have taken steps to reduce costs and improve efficiencies. These steps include structural changes in the organization to centralize common functions while maintaining the desired focused factory approach to roll label, graphic, distribution and technical markets. The Six Sigma initiatives will be an additional source of cost savings and have established the framework to improve quality in the manufacturing plants.

With these continuing initiatives to improve our business and the anticipated market improvements, we are well positioned in all respects to meet the growing needs of our customers as well as the performance expectations of our shareholders."

Earnings Outlook

Based upon our current forecasts, we expect third quarter results to be about 10% to 12% higher than prior year quarterly levels, with full year diluted earnings in the \$2.55 to \$2.65 per share range as previously stated.

Bemis Company, Inc. will Webcast an investor telephone conference regarding its second quarter 2001 financial results this morning at 11 a.m., Eastern Daylight Time. Individuals may listen to the call on the Internet at *www.bemis.com* under "Investor Relations". However, they are urged to check the website ahead of time to ensure their computers are configured for the audio stream. Instructions for obtaining the required, free, downloadable software have been posted on the site.

Bemis Company is a major supplier of flexible packaging and pressure sensitive materials used by leading food, consumer products, manufacturing, and other companies worldwide. More information about the company is available at our website, *www.bemis.com*.

Statements in this release that are not historical are considered "forward-looking" and are subject to certain risks and uncertainties. Actual future results and trends may differ materially from historical results or those projected in any such forward-looking statements depending on a variety of factors which are detailed in the Company's regular SEC filings including the most recently filed Form 10-K for the year ended December 31, 2000.

BEMIS COMPANY, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF INCOME

(in thousands, except per share amounts)
(unaudited)

	 Three Mo	nths Ei e 30,	ıded	 Six Mont Jun	hs End	led
	2001		2000	2001		2000
Net sales	\$ 581,569	\$	537,332	\$ 1,158,964	\$	1,049,948
Costs and expenses:						
Cost of products sold	459,245		423,267	919,453		828,330
Selling, general and administrative expenses	52,944		47,914	109,038		98,768
Research and development	2,704		2,448	5,188		5,024
Interest expense	8,244		6,383	18,925		12,106
Other costs (income), net	922		113	564		724
Minority interest in net income	121		157	220		203
	 	_		 		
Income before income taxes	57,389		57,050	105,576		104,793
Provision for income taxes	 21,900		21,700	40,400		39,800
Net income	\$ 35,489	\$	35,350	\$ 65,176	\$	64,993

Basic earnings per share of common stock	\$.67	\$.67	\$ 1.23	\$ 1.22
Diluted earnings per share of common stock	\$.67	\$.66	\$ 1.23	\$ 1.21
Cash dividends paid	\$.25	\$.24	\$.50	\$.48
Weighted average common shares and common stock equivalents outstanding	52,922	53,617	52,940	53,652

BEMIS COMPANY, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEET

(dollars in thousands) (unaudited)

	Jun. 30, 2001		Dec. 31, 2000
ASSETS			
Cash	\$ 34,892	2 \$	28,910
Accounts receivable-net	293,165		301,974
Inventories	284,427		274,323
Prepaid expenses	30,990		34,752
Total current assets	643,474		639,959
Property and equipment, net	830,214		825,754
Goodwill	290,139)	297,898
Intangible assets, deferred charges, and other assets	122,515		125,032
8,	,		- ,
Total	412,654	, 	422,930
TOTAL ASSETS	\$ 1,886,342	\$	1,888,643
LIABILITIES AND STOCKH	OLDERS' EQUITY		
Current portion of long-term debt	\$ 178,193	\$	227,459
Short-term borrowings	3,766		7,353
Accounts payable	205,693		207,115
Accrued salaries and wages	36,589		43,661
Accrued income and other taxes	20,376		9,509
Total current liabilities	444,615		495,097
Long-term debt, less current portion	439,133		437,952
Deferred taxes	108,533		103,621
Other liabilities and deferred credits	59,090		51,646

Total liabilities	1,051,373	1,088,316
Minority interest	1,833	1,570
Stockholders' equity:		
Common stock (61,195,954 and 60,972,802 shares)	6,119	6,097
Capital in excess of par value	242,335	237,100
Retained income	893,277	854,506
Other comprehensive income (loss)	(59,504)	(49,855)
Treasury common stock (8,370,388 and 8,370,388 shares)	(249,091)	(249,091)
Total stockholders' equity	833,136	798,757
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	\$ 1,886,342	\$ 1,888,643

BEMIS COMPANY, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF CASH FLOWS

(in thousands) (unaudited)

Six Months Ended

		June	e 30,	
		2001		2000
Cash flows from operating activities				
Net income	\$	65,176	\$	64,993
Adjustments to reconcile net income to net cash provided by operating activities:				
Depreciation and amortization		63,332		52,601
Minority interest in net income		220		203
Deferred income taxes, non-current portion		5,000		1,587
Losses of unconsolidated affiliated companies		1,961		989
Tax benefits related to stock incentive programs		1,400		841
Loss (gain) on sale of property and equipment		320		(15)
Changes in working capital, net of effects of acquisitions and dispositions		7,335		(34,865)
Net change in deferred charges and credits		(819)		(3,401)
	_		_	
Net cash provided by operating activities		143,925		82,933
	_		_	
Cash flows from investing activities				
Additions to property and equipment		(60,341)		(51,890)
Business acquisitions		(605)		(3,277)
Proceeds from sale of property and equipment		414		247
Other		0		10
Net cash used in investing activities		(60,532)		(54,910)

Cash flows from financing activities		
Change in long-term debt excluding debt assumed in business acquisition	1,513	37,530
Change in short-term debt	(52,501)	(1,707)
Cash dividends paid	(26,405)	(25,624)
Common stock purchased for the treasury	0	(25,065)
Stock incentive programs	(1,400)	(841)
Net cash used by financing activities	(78,793)	(15,707)
Effect of exchange rates on cash	1,382	(1,791)
Net increase in cash	5,982	10,525
Cash balance at beginning of year	28,910	18,187
Cash balance at end of period	\$ 34,892	\$ 28,712

QuickLinks

BEMIS COMPANY, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF INCOME (in thousands, except per share amounts) (unaudited)

BEMIS COMPANY, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEET (dollars in thousands) (unaudited)
BEMIS COMPANY, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF CASH FLOWS (in thousands) (unaudited)